Do university technology transfers increase firms' innovation?

by

María García-Vega and Óscar Vicente-Chirivella*

Abstract

We investigate how technology transfers from universities to private firms influence firm innovativeness. Using data on R&D acquisitions from universities of more than 10,000 Spanish firms for the period 2005-2013 and applying propensity score matching techniques, we find that technology transfers from universities strongly increase firm innovativeness. We next explore heterogeneous effects in order to analyse whether these gains are mediated by firm size and the business cycle. Our results suggest that the contribution of universities to firm innovation is particularly important for small firms and during the whole business cycle. The contribution of universities goes beyond its direct effect on innovation: We find that technology transfers from universities generate positive spillovers and enhance firms' internal R&D capabilities. Our results suggest that the knowledge generated by universities makes an important contribution to economic growth through technology transfers, which makes firms more innovative. Hence, knowledge creation by universities provides an important public good.

Keywords: Universities, Technology Transfers, Innovation, Firms

JEL classification: L25, D22, L24, O31

^{*} García-Vega: School of Economics, University of Nottingham, University Park, Nottingham NG7 2RD, United Kingdom and GEP (email: maria.garcia-vega@nottingham.ac.uk); Vicente-Chirivella: Department of Economic Analysis, Universitat de València, Avd de los Naranjos s/n, Valencia 46022, Spain. (email: oscar.vicente@uv.es).

We would like to thank Loles Añón, Simon Gächter and Joel Stiebale for their helpful comments.

1. Introduction

Do universities provide benefits to society beyond providing higher education to the young generation? This question has been at the core of the public and political debate about the role of university in society (Veugelers, 2016). One way universities can benefit society beyond education is through the transfer of their scientific research to firms which, in turn, can enhance innovation and thereby long-run economic growth (Mansfield, 1991). A core role of universities is to generate basic knowledge at the frontier of research, which is difficult to obtain through private markets.¹ Therefore, companies have incentives to acquire some research from universities (contractual technology transfers) instead of producing it themselves to remain competitive and to increase efficiency. While there is a large amount of literature on the effects of in-house R&D on innovation, and on the productivity of technology transfers from the perspective of universities,² few studies analyse the effects of contractual university technology transfers on firm innovation. In this paper, we contribute to filling this gap.

We investigate the effect of knowledge transfers from universities to private firms on firm innovativeness. A fundamental feature of universities is that they generate basic and applied research in an interlinked way. As a consequence, a large variety of different firms can benefit from university knowledge. Some small and medium sized firms lack capabilities and skilled personnel to implement incremental product innovations already known in the market.³ Some start-ups hire research university services to create and organize their own laboratories. Large firms often have incentives to develop new products and processes to stay ahead of their competitors and to reduce their costs. These large firms also often acquire basic research from universities to obtain radical innovations.⁴ Therefore studying

¹ Basic research is a public good and therefore there is often no market for creating that type of knowledge (Stephan, 1996; Lach et al., 2017).

² For example, Siegel et al. (2003a), Siegel et al. (2003b), Siegel et al. (2004), Chapple et al. (2005), Siegel et al. (2007), Macho-Stadler et al. (2007), Belenzon and Schankerman (2009) and Caldera and Debande (2010) study the performance of university technology transfer offices.

³ For instance, strawberry farmers hire agricultural engineers and chemistry services from universities in order to delay expiry dates. This is a known technology in the agricultural industry, but it is difficult to implement by small farmers. In this line, McGuik et al. (2015) show that small firms are particularly sensitive to innovative human capital in order to innovate.

⁴ For example, banks hire R&D services from computer science departments at universities in order to develop customized banking based on eye tracking technology. An example provided by Azoulay et al. (2018) and Novartis (2017) is the pharmaceutical company Novartis, which funded research on gene mutation performed at the

university technology transfers helps us to understand the economic returns to research performed at universities not only for large firms, but also for small firms, which have an important impact on local communities because, for instance, they hire predominantly from their local area.

Beyond the challenges of observing contractual technology transfers from universities to private firms from the perspective of the firms, there are selection problems, which make it hard to causally identify the effects of technology transfers. Our econometric analysis uses panel data of more than 10,000 Spanish firms for the period 2005-2013. Its panel structure permits us to treat potential selection issues and endogeneity problems. Our data contain unique information of firm acquisitions of R&D from universities. With this information, we can identify contractual technology transfers from universities to private firms.⁵ To our knowledge, this dataset is the most detailed panel database worldwide for contractual technology transfers from universities and, therefore, particularly suitable for our research purposes.

Our empirical approach is a combination of matching techniques and DiD estimations. We implement a Conditional difference-in-differences (CDiD) estimator. As a robustness check, we also perform instrumental variable (IV) regressions. We find that firms with technology transfers from universities strongly increase their innovativeness compared to firms without technology transfers. We also find a positive impact of technology transfers from universities on firm innovation by comparing knowledge transfers from universities with technology transfers from other providers, such as private firms or non-university research institutions.

The effects of technology transfers on innovation we uncover are particularly sizeable for small and medium sized firms. The distinction between small and large firms is interesting because innovation by small firms is key for productivity and for reduction of inequalities (OECD, 2018). Moreover, we find

University of Pennsylvania in order to develop immunocellular therapy against cancer. Bercovitz and Feldman (2007) study the type of R&D that is performed when large multinationals collaborate with universities.

⁵ Medda et al. (2005) and Vega-Jurado et al. (2017) use a similar characterization of university technology transfers. See Perkmann and Walsh (2007) for a discussion of different types of knowledge relationships between universities and private firms.

that the positive impact of technology transfers occurs over the whole business cycle but particularly in less financially constrained periods.

A deeper look at the data suggests that the impact of technology transfers from universities goes beyond its direct effect on innovation. Another contribution of our paper is to show that universities generate positive spillovers on patenting in regions and sectors with high concentration of technology transfers. This suggests that our direct effects are, indeed, a lower bound for the contribution of universities on firm innovation. Our final contribution is evidence that the impact of technology transfers from universities is particularly important for firms operating in R&D intensive sectors and that technology transfers also enhance firms' internal capabilities and their internal R&D resources, which implies that knowledge transfers are complementary to internal research.

This paper proceeds as follows: Section 2 discusses the related literature and the theoretical considerations. Section 3 presents the data and the description of the main variables; Section 4 discusses our econometric specification; Section 5 contains our main results; Section 6 shows additional empirical evidence, investigating heterogeneous effects, spillovers and crowding-out effects. Section 7 discusses our results and conclusions.

2. Related literature and theoretical considerations

We might expect that technology transfers from universities increase firm innovativeness for several reasons. First, there are some innovations that firms cannot develop on their own because of the lack of basic knowledge. A prominent example provided by Mansfield (1991) is the case of silicon. The study of the organic compounds of silicon by Kipping and Staudinger at the beginning of the 20th century was essential for the development of industrial silicon. The research was motivated by the academic curiosity to find how similar silicon was to carbon and not by any commercial purpose.⁶

⁶ Working at the University of Nottingham, Kipping published more than 57 research papers between 1899 and 1944. In 1904, Kipping called silicon polymers "sticky messes of no particular use". Silicon is now used in a broad range of sectors from pharmaceutical, electronics or automobile industry. During WWII, silicon products were used for aircraft engine isolation, which allowed allied fighter planes to fly at high altitudes and in low temperatures (Thomas, 2010).

Second, even if firms are aware of the basic knowledge, they might lack the technical capabilities to develop the innovations. Therefore, as Mansfield argues, in a practical sense, these innovations cannot be developed without academic research. For instance, firms rent university laboratories to perform experiments because of the high fixed costs of setting up a lab. Third, early-stage innovations require creativity, which is difficult in large firms due to their organizational bureaucracy (Williamson, 1985, Ch 6) and the routinization of their R&D investments (Baumol, 2002). As a consequence, it has been traditionally more likely that breakthrough innovations are generated in universities and then transferred to firms than being created directly in large firms.

An alternative possibility to a positive relationship between university technology transfers and firm innovation is that firms have little use of the research generated by universities, either because it is too complex or because it requires high absorptive costs. For example, firms might find that scientific knowledge is too abstract and, therefore, too disconnected from their own direct commercial purposes. Over time, research at universities is becoming more complex because more learning is required to reach and push the technological frontier (Bloom et al., 2017). Similarly, in order to use the knowledge generated by universities, firms might need to increase their investments in absorptive capacities. For some firms, particularly small firms, the costs of hiring researchers or increasing the number of researchers in their existing laboratories might be too high. Therefore, firms might prefer to focus their attention on technologies generated by close competitors or suppliers, which might be less costly to assimilate (Bikard, 2018). Moreover, these arguments suggest that there can be heterogeneous effects that mediate the relationship between technology transfers from universities and firm innovativeness. By studying a large range of heterogeneous effects, our paper contributes to a better understanding of how knowledge from universities is absorbed by private firms.

According to Perkmann and Walsh (2007) there are two main types of high-level relationships between universities and firms that can induce technology transfers: R&D contracting and cooperation. The literature suggests that these two types of relationships provide different knowledge to firms (Cassiman et al., 2010; Lucena, 2011; Vega-Jurado et al. 2017), and they involve different costs. R&D collaborations imply sharing not only codified knowledge, that is, explicit and more articulated knowledge, but also tacit sources of knowledge (Lane and Lubatkin, 1998; Almeida et al., 2003).⁷ In the case of R&D contracts, typically, only codified knowledge is exchanged and, therefore, it can be more limited than cooperation. However, an advantage of contractual R&D is that it is more focused on improving problem-solving capabilities in the innovation process than on R&D cooperation. This can have important effects on firm innovativeness because problem-solving capacities expedite experiments, enhance measurements in the labs and improve the interpretation of results (Dasgupta and David 1994; Antonelli 1999). Moreover, R&D cooperation is often not used by small firms. For example, Miotti and Sachwald (2003), Negassi (2004) and López (2008), among others, find a positive relationship between firm size and the probability of R&D cooperation. Our results suggest that R&D contracting is negatively correlated with firm size. This seems to indicate that small firms are more likely to contract R&D than be involved in R&D cooperation agreements. One possible reason is that R&D cooperation requires more highly-skilled R&D personnel and research managers than R&D contracting due to the more tacit knowledge that is being transferred.⁸ Another advantage of contractual R&D, instead of cooperation or more general measures such as sources of information or benefits from interactions with universities, is that we can account for the direction of the technology transfer, which facilitates identification.

How does our paper relate to the existing literature? By using a unique dataset on detailed contractual R&D from universities, we contribute to the literature on the effects of public research on industrial innovativeness. In his seminal paper, Mansfield (1991) used survey data from top R&D executives from major American firms to study the effects of academic research on firms' innovation performance. In order to measure academic research, Mansfield considered a question about the innovation that could not have been developed (without substantial delay) in the absence of academic research. Using the same variable as Mansfield (1991), Beise and Stahl (1999) analyze the impact of publicly financed funds on firm innovativeness using survey data for large German corporations. In

⁷ Yu and Lee (2017), Szücs (2018), Kafouros et al. (2015), Lööf and Broström (2008), Grimaldi (2002), Monjon and Waelbroeck (2003), among others study the effects of university cooperation on innovation outcomes.

⁸ For example, Lucena (2011) shows for the case of R&D acquisitions from different providers that a significant number of firms that acquire R&D do not perform internal R&D, which suggests that learning from R&D acquisitions is possible even without the adoption of in-house R&D.

both studies, university research has an important positive effect on firm innovation. In our study, we also find a positive effect of universities on firm innovativeness. However, in contrast to these studies, our paper focuses on the impact of contractual university technology transfers and we provide new evidence regarding heterogeneous effects. Another difference between our paper and these studies is that our data integrate large and small firms and, therefore, we are able to identify the contribution of universities also for medium and small sized firms.

Several studies use cross-section survey data to study the influence of university technology transfers on firm innovativeness. For example, Cohen et al. (2002) study how university public research influences industrial R&D as compared to other sources of information. Their study suggests that the effect of university research varies across industries, the pharmaceutical industry being one of the most positively affected industries. Arvanitis et al. (2008) use Swiss survey data to study how different types of university-firm relationships influence firm innovation. Their findings suggest that university research knowledge is very important for firms' sales of new products. Bishop et al. (2011) study how firms' interactions with universities enhance different types of research outputs such as problem solving, generation of patents or improvement of the firms' understanding. In contrast to Arvanitis et al. (2008) and Bishop et al. (2011) we use panel data, which allow us to address the potential selection bias problem, and we can also provide detailed determinants for firms to obtain technology transfers. Furthermore, we provide detailed evidence about contractual R&D and we discuss the intensity of technology transfers.

Our paper is most closely related to Medda et al. (2005) and Vega-Jurado et al. (2017). Medda et al. (2005) study the effect of R&D acquisitions from universities and other types of external R&D on firm productivity. They do not find a statistically significant effect on firm productivity. One possible reason for this result, as they explain in the paper, is that contractual R&D from universities might require a longer time frame to influence firm productivity than the one that they consider. In contrast, our interest is on firm innovativeness. A further novelty of our analysis is that we provide evidence on indirect effects of technology transfers testing whether they have an impact upon firms' internal capabilities and spillovers.

Vega-Jurado et al. (2017) investigate the effects of both contractual and cooperative relationships between universities and private firms on different types of product innovations. They show that contractual R&D influences incremental product innovation, while cooperation increases radical product innovation. In our study, we also find a positive relationship between contractual R&D and incremental product innovation. In contrast to them, we also investigate the effects of contractual R&D on process innovation and patents, in addition to product innovation. Another difference with Vega-Jurado et al. (2017) is that we use, among other techniques, instrumental variable procedures and matching approaches in order to control for selection. Moreover, differently from them, we compare the impact of technology transfers from universities with respect to technology transfer coming from private firms or non-university research institutions. A recent paper by Fudickar and Hottenrott (2019) investigates the impact of formal and informal interactions with universities on the innovativeness of German startups. They combine cooperative and contractual research (among other relationships) to construct their indicator of formal interactions. Differently from Fudickar and Hottenrott (2019), we focus specifically on contractual R&D. Finally, we shed light on the effects of technology transfers for the average firm and we distinguish by sectors, firm size and different macroeconomic periods.

3. Data and description of the main variables

In this section we describe the dataset and the main variables that we use for our empirical analysis. Further details are in the following sections and in Tables 1 and 2 where we present descriptive statistics and definitions of the main variables. Our goal is to analyze the effect of technology transfers from universities on firms' innovation. For this purpose, we use a dataset that comes from a survey of Spanish firms called Panel de Innovación Tecnológica (PITEC) for the period 2005-2013. PITEC represents Spain's contribution to the Europe-wide Community Innovation Survey (CIS). It is the result of the collaboration between the Spanish National Statistics Institute, the Spanish Science and Technology Foundation and the Foundation for Technological Innovation.⁹ PITEC is an unique dataset that includes

⁹ PITEC applies the methodological rules defined in the *Oslo Manual* OECD's (2005a). Details on PITEC and data access guidelines can be obtained at: <u>https://icono.fecyt.es/pitec</u>.

a representative sample of the universe of Spanish firms. The dataset contains detailed firm-level information on a number of firm characteristics such as number of employees and turnover and different measures of innovation inputs and outputs.

Our sample is an unbalanced longitudinal panel. We consider nine waves from the PITEC dataset from 2005 to 2013. On average, there are 10,760 firms during our sample period and, among these, there are 10,042 firms with four consecutive observations and 8,589 firms with nine consecutive observations. We show the distribution of firms across the different year-waves in the first column of Table 1.

3.1. Main independent variable: Technology transfers from universities

We are interested in the effects of technology transfers from universities upon firm innovativeness. Our measure of technology transfers are R&D services acquired from Spanish universities by firms operating in Spain. In the survey, each company indicates its R&D acquisitions, that is, its purchases of R&D services.¹⁰ R&D acquisitions are defined in the survey as:

"Acquisitions of R&D services outside the firm through contracts, informal agreements, etc... Funds to finance other companies, research associations, etc... that do not directly imply purchases of R&D services are excluded".

With this information, we construct the variable *university technology transfers*, which is a dummy variable that takes the value one if a firm has expenditures in R&D services from Spanish universities and zero otherwise. Measures similar to our measure of technology transfers from universities are used by Fudickar and Hottenrott (2019), Vega-Jurado et al. (2017) and Medda et al. (2005). Tests, technological support, researchers or faculty consulting are some examples of the type of R&D services that companies acquire from universities and that are embedded in our measure of technology transfers.

The advantage of our measure with respect to other measures is twofold. First, it captures an intensive type of knowledge transfer from universities to companies, which is difficult to obtain using

¹⁰ R&D services are defined in the survey as: "Creative work to increase the volume of knowledge and to create new or improved products and processes (including the development of software)".

only measures of patent citations or licensing (D'Este and Patel, 2007).¹¹ For example, Cosh et al. (2006), using a survey of UK and US firms, report that firms consider that the most important types of university-industry interactions contributing to their innovation activities are testing and standards, problem-solving, and innovation expenditures to universities. Second, R&D acquisitions are largely used by both large and small firms, while other measures such as cooperation are not used so often by small firms as they require highly-skilled R&D personnel and research managers (Teirlinck and Spithoven, 2013). Since in our dataset there is a large number of small firms, it is likely that our estimations would suffer from a strong selection bias if we use cooperation as our measure of technology transfers because we would not be accounting for technology transfers for small firms. The disadvantage of R&D acquisitions from universities is that informal contacts between firms and universities are not included. Since many of these informal contacts are important for firm innovation and they are likely to precede the time of the formal R&D acquisition from universities, our results can be considered as a downward biased estimation of the effects of university technology transfers on firm innovation.

In our sample, the percentage of firms with technology transfers from universities is 6.8% (see Table 2). Moreover, 8.14% of firms obtain technology transfers more than twice during the sample period, although this number is reduced to 0.58% for firms with continuous technology transfers for the whole period. In Table 1, we report the number of firms with technology transfers from universities by year (column 2). We also present information on the number of firms that, in our sample period, start obtaining technology transfers in a given year for the first time (column 3), on firms that never have technology transfers (column 4) and on overlapping firms, i.e., firms that in a given year do not have technology transfers but had them in previous years or will have them in the future (column 5). On average, in each year, 1.5% of firms start acquiring technology from universities for the first time. Additionally, on average, 13.4% of observations correspond to firms that do not obtain technology

¹¹ One disadvantage of patents citations as a measure of technology transfers is that patenting suffers from a double skewed phenomenon. Almost 40 per cent of all university patents around the world are held by 50 institutions. Moreover, within these 50 institutions, the large majority are from either the US or the UK (Veugelers, 2016). Some studies have used licenses or royalties as measures of technology transfers. However, licensing is even more concentrated than patents. For the UK, one-third of the total income generated by licenses is concentrated in two licensors (Russell Group, 2010). Scherer and Harhoff (2000) show that 93 per cent of royalties received on inventions in 1991 were held by six research-oriented US universities.

transfers, but that either in previous years have had technology transfers or they will have them in the future. These figures indicate the importance of technology transfers from universities.

In Table OA1 in the on-line Appendix, we present the number of observations and percentage of firms with technology transfers by sectors of activity. Transfers from universities are highly concentrated in R&D-intensive sectors such as "Pharmaceutical", "Chemicals" and "R&D services" but also in the "Agricultural" sector, which is an important industry in the Spanish economy.

3.2. Dependent variables

Our dependent variables are measures of innovation output at the firm level. In particular, we consider three different measures of firm innovativeness in our baseline specifications: having product innovation, having process innovation and having patents.¹² In the robustness checks section we include additional indicators of innovation outputs. Product innovation, process innovation and patents are well-established indicators of innovation used in a large number of empirical studies.¹³ We measure product (process) innovation, as a dummy variable that takes the value one if the firm reports having introduced new or significantly improved products (processes) in the current or previous two years. In the same vein, patents is a dummy variable that takes the value one if a firm reports having patents in the current or previous two years and zero otherwise.

The advantage of these measures of innovativeness is that they directly refer to the output in the context of a knowledge production function, in which technology transfer is an input. The distinction between product and process innovations allows us to differentiate between demand-based innovations (product innovations) and cost-reduction innovations (process innovations). Patents provide a good signal of the degree of novelty of firm innovativeness. Moreover, since patents are also derived from administrative data, they are likely to be more objective than other indicators of innovation output (Haucap et al., 2019). In fact, patents have been widely used in recent studies to measure innovation output (Aghion et al., 2009; 2013;

¹² See Mairesse and Mohnen (2005) for a detailed explanation of how CIS surveys are structured and the main innovation indicators in this type of survey.

¹³ See Geroski et al. (1997), Griffith et al. (2006), Cefis and Orsenigo (2001), Cefis (2003), Martínez-Ros and Labeaga (2009), Clausen et al. (2011), Tavassoli and Karlsson (2015) or Ganter and Hecker (2013), among others.

Bena and Li, 2014; Seru, 2014; Haucap et al., 2019).

Table 2 shows descriptive statistics of the main variables for the whole sample and differentiating by technology transfers from universities' status. Firms with technology transfers from universities are characterized by a higher innovation profile than those without technology transfers. The percentage of firms having introduced either a product (process) innovation or a patent is higher for companies with technology transfers from universities than without technology transfers. The largest difference is for the variable product innovation. More than 70% of firms with technology transfers reported at least a product innovation during the current or previous two years while this percentage is less than 50% for companies without technology transfers. The differences in process innovation and patents are also higher than 20 percentage points. Moreover, there are also important differences on the sales from products new to the market (firm). Table 2 further shows a higher level of human capital for firms with technology transfers from universities. These results suggest that there is a positive correlation between technology transfers from universities and innovation. In the following sections, we measure these effects controlling for selection.

4. Econometric specification

We aim to study the effect of university technology transfers on firm innovativeness. To face this objective, in our main specifications, we estimate two empirical models that combine propensity score matching with difference-in-differences (DID) estimators.

The first model considers changes that happen within the same firm by examining companies that *start* for the first time to obtain technology transfers from universities. This approach allows us to determine the average treatment effect on the treated (ATT), which is the difference between the

innovation outcome variable of firms with technology transfers from universities and their innovation outcome without technology transfers.¹⁴ The ATT can be specified as follows:

$$ATT = E[y_{t+1}^1 | T_t = 1] - E[y_{t+1}^0 | T_t = 1]$$
(1)

In the expression above, the term y_{t+1}^1 is the innovation outcome in case of technology transfers from universities, y_{t+1}^0 is the innovation outcome without technology transfers from universities, and T_t is a dummy variable that takes the value one when there are technology transfers from universities. The evaluation problem is that the counterfactual outcome of not having technology transfers is unobserved for the treated firms.

The matching technique allows us to find a set of firms with the same observable characteristics as the treated group before having technology transfers but that did not receive the treatment. We employ a one-to-one propensity score matching within the same year with replacement (Rosenbaum and Rubin, 1983, Rosenbaum and Rubin, 1985). In each year, for each firm that will start for the first time obtaining technology transfers the following period, we identify a control firm with similar observable characteristics that never obtained transfers from universities during our sample period. In this way, we ensure that there is no overlap between treated and control group after matching. The matching procedure controls for observable firm characteristics that can influence both the probability of having technology transfers and innovating by considering a comparable sample of firms. Our identifying assumption is that, conditional on the observable characteristics that are relevant for technology transfers. In other words, we assume that, in the absence of technology transfers, the outcome of the treated group would not have been systematically different than the outcome of the control group. The DiD estimator measures the changes to innovation outcome between pre- and post-technology transfers for the treated versus the control group and, therefore, controls for time-invariant unobservable characteristics.

¹⁴ See for example Guadalupe et al. (2012), Haucap et al. (2019), Jabbour et al. (2019), Javorcik and Poelhekke (2017), among others.

The propensity score that we use for the matching procedure comes from a Probit model where we calculate the probability of having technology transfers from universities on a set of observable firm characteristics, denoted by X_{it-1} . Formally:

$$T_{it} = \begin{cases} 1 & \text{if } \gamma + X'_{it-1}\rho + d_t + \xi_{it} > 0\\ 0 & \text{if } \gamma + X'_{it-1}\rho + d_t + \xi_{it} \le 0 \end{cases}$$
(2)

In equation (2), the vector X_{it-1} reflects pre-treatment firm characteristics that influence the likelihood to have technology transfers from universities, d_t denotes time dummies, and ξ_{it} is the error term, which we assume is normally distributed with variance σ_z^2 . In all regressions, we use cluster robust standard errors at the firm level. We also control for time-specific sectoral shocks to the economy that might affect technology transfers. After we estimate the propensity score from equation (2), we pair each treated firm with the closest untreated firm by caliper matching with replacement and we obtain our DiD estimator as follows:

$$y_{it+2} = \alpha + \beta T_{it} + \varepsilon_{it}, \tag{3}$$

where y_{it+2} denotes firm innovation output and β is the DiD parameter of interest and it measures the ATT effect.¹⁵

Our approach, matching within the same year and considering as treated group firms that start obtaining technology transfers for the first time and considering the control group firms that never had technology transfers, has the advantage that it avoids the potential problem of overlapping treated and control firms. Moreover, it ensures that we control for time-specific confounding factors that affect firms with and without technology transfers. However, given that some treated firms receive the treatment more than once and that our panel is relatively short, it is possible that the composition of treated and control groups might change over time due to some unobservable time-variant characteristics. For

¹⁵ The innovation output variables are included with a two-period lead. That is, we study the probability of having innovations up to two years after receiving technology transfers from universities. The reason for the two-year lead is due to the definition of the variables in the survey. Following the usual definitions in Community Innovation Surveys, in our dataset, innovation output questions are for the current and previous two years, while innovation inputs and accounting variables are for the current period.

example, at the beginning of our sample period, some of the firms in the control group might have received a treatment in the near pre-sample period, or selection into the treatment might be influenced by unobserved reasons that are not year or firm-specific but university time-variant-specific. In order to address potential concerns regarding the composition of the groups over time, we estimate a conditional difference-in-differences estimator (CDiD) with repeated cross-sections (Blundell and Costa-Dias, 2002; Aerts and Schmidt, 2008). This model allows to control for possible changes in the composition of the treated and control groups that change over time.

We estimate the average treatment effect on the treated firms using the CDiD methodology employed by Aerts and Schmidt (2008), which they apply to the effects of additionality of R&D subsidies. The approach by Aerts and Schmidt (2008) considers three different matchings. Let us call t_0 and t_1 two consecutive time periods. The first match is in period t_1 between a treated firm, denoted by *i* that receives the treatment in period t_1 and a non-treated firm, denoted by *h* that does not receive the treatment in period t_1 . The second match is found in period t_0 between firm *i* and a non-treated firm in period t_0 , which we denote by *k*. The third match is also in period t_0 , in this case, it is between firm *h* and a non-treated firm in period t_0 , which we denote by *j*. The average treatment effect on those treated is calculated as follows:

$$ATT^{CDiD} = \left(E[y_{i,t_1}^1 | T_{i,t_1} = 1] - E[y_{k,t_0}^0 | T_{k,t_0} = 0] \right) - \left(E[y_{h,t_1}^0 | T_{h,t_1} = 0] - E[y_{j,t_0}^0 | T_{j,t_0} = 0] \right)$$
(4)

An important assumption for our identification strategy is that technology transfers from universities do not have an indirect effect through spillovers into the control groups, as this would be a violation of the Stable Unit Treatment Value Assumption (SUTVA assumption). We initially rule out this possibility but investigate this assumption in Section 6.

5. Effect of technology transfers from universities on firm innovativeness

In this section we present evidence regarding the effect of technology transfers from universities on firm innovativeness. First, we estimate this relationship for the whole sample. Second, we estimate the impact for the matched sample using starters as treated firms and firms that never obtain technology transfers as control group. Third, we estimate the average treatment effect with the CDiD methodology and we show robustness checks including our IV specification. Fourth, we exclude from our sample firms without any type of technology transfer. In this way, we compare the effect of technology transfers from universities with the impact of technology transfers from other providers and assess its relative importance. For this analysis, we also use the CDiD methodology.

5.1. Results from the whole sample

Before we report our results from the matching samples, we first show evidence based on the whole sample without controlling for the potential selection bias or endogeneity issues. We present the results in Table 3. In panel A, the dependent variable is product innovation; in panel B, the dependent variable is process innovation; and, in panel C, the dependent variable is patents.

We report estimates including different controls and firm fixed effects. In columns 1 and 2, we do not include firm fixed effects. From column 3 to 5, we add firm fixed effects. From column 2 to 5, we include lagged control variables. In column 3, we include firm fixed effects using the Wooldridge (2005) correction methodology.¹⁶ Following this method, the unobserved individual effect (α_i) is conditioned on the initial values of the dependent variable (y_{i0}) and the individual mean of the time-varying covariates (\bar{x}_i), allowing for correlation between the individual effect and the observed characteristics. In columns 1, 2 and 4, we control for sector fixed effects while in column 5 we include sector-time fixed effects. In all regressions in all panels, we include year fixed effects. All standard errors are clustered at the firm level.

In all columns, and in all panels, we show that university technology transfers are always strongly positively related to any type of innovation output. For example, the estimated coefficient of university technology transfers in column 1a suggests that having technology transfers increases the likelihood of having product innovation by 24.2 percentage points. Once we include firm fixed effects to control for

¹⁶ This methodology allows the individual effect to be correlated with the regressors and solve the 'initial conditions problem'. The initial conditions problem arises when the first observation for each firm in a panel does not coincide with the first year of this firm; that is, when we do not have information about firms from the very beginning. Since the first observation for each firm is affected by the same process that will affect the variable from the first year of the observation period, this variable would be endogenous.

time invariant firm characteristics in columns 3a to 5a, we find that this effect remains positive and highly significant, but the magnitude is lower than in previous specifications. In particular, the estimated coefficient in column 5a suggests that having technology transfers from universities might increase the likelihood of having product innovations by 2.8 percentage points. The estimated coefficients in panel B for process innovation are of similar magnitude to those for product innovation in panel A. In the most conservative estimations, in columns 4b or 5b, we observe that university technology transfers increase process innovation by 2.3 percentage points. Finally, the results in panel C, in the most conservative estimations, indicate that having university technology transfers increases the likelihood to patent by 1.9 percentage points (column 5c).

5.2. Results using starters

Before turning to the effect of technology transfers from universities on firm innovativeness with the matched samples, we first summarize the estimates of the probability model that we use to obtain the propensity scores for our matching procedure. Our dependent variable is a dummy variable that takes the value one when there are technology transfers from universities. As control variables, we follow Cassiman and Veugelers (2006), Piga and Vivarelli (2004) and Parmigiani (2007) to consider determinants of external knowledge acquisition. We include measures of internal R&D in the regressions (measured as the natural logarithm of a firm's intramural R&D expenditures, and the natural logarithm of the number of employees working in R&D), which also control for the level of absorptive capacity of the firm. We control for firm size (with the natural logarithm of the total number of employees, and the natural logarithm of the physical investments) to account for economies of scope. We include firm exporting status and an indicator that takes the value one if the firm belongs to a business group in order to control for firm internationalization. We add product, process and patent dummy variables, as well as time and industry dummies. To avoid reverse causality problems, we lag our explanatory variables one period.

The results from the probit specification for starter firms and untreated firms that never receive the treatment during the sample period is reported in column 1 of Table A1 in the Appendix. Our estimates suggest that firms that have patents are likely to obtain technology transfers from universities in the following period. Moreover, smaller firms (in terms of employment) but with more investments in

physical capital seem more likely to obtain technology transfers from universities. With respect to R&D inputs, firms with more researchers in R&D and more internal R&D expenditures are more likely to obtain technology transfers. This suggests that absorptive capacity is important in order to obtain technology transfers from universities; this is in line with Cassiman and Veugelers (2006) who find complementarity between internal and external knowledge.

Based on the results from equation (2), within the same year, we pair by caliper matching with replacement (with caliper equal to 0.0001) each treated firm with the closest untreated firm that never received the treatment.¹⁷ We report the descriptive statistics of the matched sample in Table OA2 in the on-line Appendix. The matching procedure works well. In Table OA3 in the on-line Appendix, we report balancing tests after matching. When we compare the sample means of the variables used in the matching procedure, we find that there is no statistically significant difference in the pre-acquisition period. Our matching specification generates well-balanced samples, which implies that control (754 firms without technology transfers) and treatment groups (769 firms with technology transfers) are equivalent in their overall observable characteristics before treatment.

In Table 4, we present the ATT effect of technology transfers from universities on firm innovation after matching. In column (1) we report the estimate for product innovation; in column (2) we present the result for process innovation; and, finally, in column (3) we show the estimated coefficient for patents. In all columns, the estimates indicate a positive and statistically significant effect of technology transfers from universities on innovation outputs. The results suggest that having technology transfers from universities increases product innovations by 7.8 percentage points, process innovations by 6.9 percentage points and patenting by 9.3 percentage points.

In the last three columns of Table 4, we investigate how the impact of technology transfers from universities on firm innovation varies over different intensities of technology transfers. For this purpose,

¹⁷ We use a one-to-one caliper matching with replacement, such that each starter firm, the year before it starts to obtain technology transfers from universities is matched with a firm that never obtains technology transfers. We decide for a matching with replacement in order to minimize the bias of the matching and maximize the number of observations in the final sample. As Stiebale and Woessner (2019) explain, the choice between with or without replacement involves a trade-off between bias and variance. Our results are also consistent with alternative matching algorithms such as caliper closest neighborhood and a reweighted estimator. These results are available upon request.

we construct four dummies that indicate the different quartiles of technology transfers intensity, where we define intensity as the ratio between technology transfers from universities over total R&D expenditures. The results show consistently for product innovation (column 4), process innovation (column 5) and patents (column 6) that as intensity increases the effect of technology transfers on innovation declines. This suggests that firms need a certain degree of absorptive capacity in order to benefit from university technology transfers. Moreover, these findings might indicate that technology transfers from universities complement the internal capabilities of the firms. We investigate this issue in more detail in Section 6.

5.3. Results using CDiD with repeated cross-section methodology

To further check the evidence presented above, we estimate the effect of technology transfers from universities using the CDiD with repeated cross-section methodology described in Section 3. We report the descriptive statistics for the different matchings in Table OA4 in the on-line Appendix. In Table OA5 in the on-line Appendix, we present the balancing test for the three different matchings required for our analysis. We show balancing tests where we aggregate across years. The matching that we employ is the nearest neighbour with replacement with caliper equal to 0.0001. The reason to use the nearest neighbour instead of one-to-one matching as in the estimations in the previous section is to avoid having to discard a large number of observations, which could reduce the estimation power (Stuart, 2010). In order to gain some insight into the general characteristics of the firms that obtain technology transfers from universities and not only of starters, in column (2) of Table A1 in the Appendix, we present the probit model that we use in our analysis. The determinants of obtaining technology transfers in column 2 are similar to those for starters in column 1. The matchings perform well across all the different variables. The comparison of sample means suggest that the different groups have statistically similar characteristics for all the observable variables. Moreover, the overall balancing tests show that the mean and median bias between groups have significantly declined after matching for the three matchings.

In Table 5, we present the ATT on the treated. The results support previous estimations. We find that technology transfers have a positive and statistically significant effect on the different measures of

firm innovativeness. The findings in Table 5 suggest that technology transfers from universities increase product innovation by 4.3 percentage points (column 1), process innovation by 7.2 percentage points (column 2) and patents by 5.5 percentage points (column 3). In the next section, we present several robustness checks of our results.

5.4. Robustness checks

We perform several sensitivity tests that we present in the Appendix, including longer pre-treatment trends, alternative definitions of our innovation output variables, a placebo test. For these three robustness checks, we use the CDiD estimation methodology.¹⁸ We also report an IV specification.

The difference-in-differences methodology is based on the assumption that the treatment and control group have statistically similar pre-treatment trends. We perform an additional test in order to control for common pre-existing trends by including two years of pre-treatment data instead of one year of the pre-treatment data. The results reported in Table A2 in the Appendix are, again, similar to those of previous specifications. Balancing tests are shown in Table OA6 in the on-line Appendix. In all cases, we observe that technology transfers from universities lead to an increase in firm product innovation and patents. This suggests that our results are not biased by longer pre-treatment trends.

We next explore the sensitivity of our results to alternative definitions of our innovation output variables. One possible concern is that our output measures in Table 5 are not properly capturing innovation output for continuous successful innovators. For example, in a given year, a very innovative company and a company with only one innovation are treated the same using dummy variables as innovation output measures. For this reason, in Table A3 in the Appendix, we present results for three continuous measures of innovation output. The first two measures capture innovative sales and are defined as the logarithm of the sales coming from products new to the market or products new to the firm, respectively, in the current or previous two years. This allows us to distinguish between radical innovations, in the case of innovations new to the market, and incremental innovations, in the case of

¹⁸ The different estimated results presented in the paper using the CDiD methodology are robust to the alternative specification of using starters.

innovations new to the firm. In addition, we include a measure of patent intensity. Our variable is the logarithm of the number of patents plus one, to deal with zeros (see for example Haucap et al., 2019 or Stiebale, 2016). The results show that the ATT effect for sales from new products to the firm and number of patents are positive and statistically significant. The effect is also positive for sales from new products to the market, but it is not significant at conventional statistical levels. This suggests that university technology transfers lead, on average, to an increase in the intensity of firm innovativeness. Moreover, these estimations confirm our previous results with respect to patents.

In order to further assess the robustness of the results presented in Table 5, we estimate a placebo regression where we assign the treatment status randomly to the control group. We present the results from the balancing test in Table OA7 in the on-line Appendix and from the ATT effect in Table A4 in the Appendix. The results from these placebo regressions are very different from previous estimations. We now find no significant differences between control and treatment groups in terms of product innovation, process innovation, and patenting.

The DiD model combined with the matching estimators described above control for time-invariant unobservable characteristics and for time-variant observable characteristics. In order to address concerns regarding the potential bias due to the omission of unobservable time-variant characteristics that could affect both transfers of technology and innovation, such as changes in managerial practices, we use an instrumental variable approach. In these specifications, we use as instrumental variable *the importance of conferences, fairs, trade shows, or exhibitions as a source of information* measured at the average of the industry and regional level and pre-sample.¹⁹ The validity of this instrument rests on the assumption that the pre-sample importance of conferences, fair trades, or exhibitions in a sector within a region can influence the technology transfers that a firm receives from universities as well as networking with university scientists, but it is exogenous to unobservable time-variant firm characteristics.²⁰ The reason is that this variable is not measured at the firm level and it precedes the years of the technology transfers.

¹⁹ This measure is constructed for the year 2004.

²⁰ See for example Siegel et al. (2004) for the importance of conference and expositions to establish relationships between business and universities and to promote technology transfers. See for example Appleyard (1996) for

We present the results in Table A5 in the Appendix. In the bottom part of the table, we show the first stage regression of the 2SLS estimations as well as the Kleibergen-Paap F-statistics. The instrument is significantly positively related to technology transfers from universities and the Kleibergen-Paap F-statistic, which is considered an approximation of the distribution of the weak-instrument yields values above 20 (the critical value for a maximum IV bias of 10% of the weak identification test is 16.38, Stock and Yogo, 2002). The IV point estimates presented in the top part of Table A5 are positive and statistically significant, which confirms the evidence presented in previous estimations. After establishing with several robustness checks that technology transfers from universities increase firm innovativeness, in the following section we study additional empirical evidence to assess the contribution of the technology transfers from universities.

5.5. Impact of technology transfers from universities on firms' innovation versus technology transfers from other providers

In order to gain further insight into the importance of the contribution of technology transfers from universities, we compare differences in innovation outputs between firms with technology transfers from universities (treatment group) and firms with technology transfers from other sources that do not include universities (control group), where technology transfers from other sources are acquisitions of R&D from other private companies or research associations (not including universities). In this way, we can assess the differential contribution of transfers from universities and transfers from other providers. If the estimated ATT effect after matching is positive (negative) and significant, it means that the contribution of technologies coming from universities is larger (smaller) than technologies from other sources. If it is not significantly different from zero, it implies that technology transfers from universities have a similar effect than technologies from other sources. This comparison is in the spirit of Medda et al. (2005) who study private returns of research projects with universities and research projects from other external sources on firm productivity for a sample of Italian firms.

types of knowledge flows in Japanese firms or Monteiro and Birkinshaw (2017) for different types of technology sourcing.

We present the estimated ATT effects using the CDiD methodology in Table 6 and the balancing test in Table OA8 in the on-line Appendix. All the estimated coefficients are positive and statistically significant at conventional levels. This suggests that the contribution of university technology transfers to product and process firm innovativeness is larger than the effect of technology transfers coming from other providers. The results suggest that technology transfers from universities increase product innovation by 13.6 percentage points; process innovation by 13.7 percentage points and patents by 12.5 percentage points more as compared with firms that obtain technology transfers coming from other providers. This confirms the importance of the contribution of universities for highly valuable innovations. We provide evidence supporting the Rosenberg and Nelson (1994) idea that universities play an important role in the development of innovations, and particularly for those that are patented.

6. Additional empirical evidence on the role of technology transfers for firm innovativeness

In this section, we first explore different heterogeneous effects differentiating firms by size and in different sample periods. Then, we study the effect of technology transfers from universities differentiating by Pavitt sectoral taxonomy. Finally, we analyse whether the contribution of technology transfers from universities goes beyond the direct effect on innovation by exploring spillover effects and the possibility of crowding-out internal R&D inputs.

6.1. Heterogeneous effects

6.1.1. Who benefits from technology transfers? SMEs vs non-SMEs

A natural question about the above estimated effects of technology transfers from universities on firm innovativeness is which particular firms benefit from the technology of the universities. In this section, we distinguish between small or medium firms (SMEs) and large firms (non-SMEs). We follow the definition of the OECD (2005b) and consider that a firm is an SME when its number of employees is less than 250. This difference is important to understand the economic contribution of universities. Small firms are fundamental for job creation, growth potential and aggregate fluctuations, as well as

local growth (Aghion et al., 2015; Audretsch et al., 1999; Autio et al., 2014; Decker et al., 2014 and Haltiwanger et al., 2013, among others). They might also be subject to financial constraints, which can reduce their possibilities to innovate and to grow (Siemer, 2019). This implies that analysing the role of technology transfers from universities distinguishing by firm size provides important information about the economic contribution of universities for economic growth.

We stratify the sample by distinguishing between SMEs and non-SMEs. We present the balancing test for SMEs and non-SMEs using the CDiD methodology in Table OA9 in the on-line Appendix. In Table 7, in columns 1, 2 and 3, we show results for SMEs and in columns 4, 5 and 6 for non-SMEs (for product innovation, process innovation and patents, respectively). Our results in Table 7 show that technology transfers are positive and statistically significant for process innovations and patents. In the rest of the specifications, technology transfers are positive but not significantly different from zero. This suggests that SMEs, particularly, profit from the technology transfers from universities in terms of process innovations and number of patents. Overall, the results suggest that the effect of technology transfers from universities on firm innovativeness is more important for SMEs than for non-SMEs.

6.1.2. Recession and non-recession periods

Our sample period includes the global financial crisis and the Great Recession of the late 2000s. The Great Recession in Spain was particularly harsh and lasted from 2008 to 2013 (Almunia et al., 2018). These were times of severe financial constraints, which allows us to study the contribution of universities to innovation during two clearly differentiated periods of the business cycle. In particular, we study the differential effect of technology transfers from universities in the recession and in the non-recession period.

Aghion et al. (2012) show that internal R&D investments are pro-cyclical when firms face tighter credit constraints (see also López-García et al., 2013 and Beneito et al., 2015). Therefore, one possibility is that when firms are financially constrained, as during the times of the Great Recession, firms tend to rely on the knowledge generated by universities instead of their own research. The reason is that innovations might be cheaper to generate with knowledge from universities than if firms have to develop

their own internal research. High sunk costs related to R&D investments jointly with the fixed costs to remain in the activity (Aw et al., 2011), plus the dramatic credit constraints suffered by firms during the recession period, may have hampered internal R&D investments. As a consequence, the effect of the technology transfers from universities on innovation might be more important during the recession period than during the non-recession period.

An alternative possibility is that the lack of finance reduces the productivity of the technology transfers from universities. For example, Mohnen and Röller (2005) show for a sample of four European countries that the lack of finance interacts with the productivity of several variables that affect innovation output such as internal R&D or regulations. Moreover, the public funding of Spanish universities fell by 27.7% during the recession period (Sacristán, 2017). This decline in public funding to universities might have negatively affected the productivity of the technology transfers from universities. Consequently, it is possible that during periods of financial constraint the contribution of the university technology declines. From an empirical point of view, this is an open question. For this reason, we next analyse whether there are significant differences between the non-recession and recession period.

We present the balancing test for the non-recession and recession period effects using the CDiD methodology in Table OA10 in the on-line Appendix. In Table 8, in columns 1, 2 and 3, we show results for the non-recession period and in columns 4, 5 and 6 for the recession period (for product innovation, process innovation and patents, respectively). In all cases, the estimated ATT is positive, with the exception of patents for the recession period. For process innovations, the estimated ATT effect is statistically significant during both the recession and non-recession period. The estimated ATT effect for patents is statistically significant for the non-recession period. This suggests that the contribution of technology from universities to firm innovativeness that reduce costs are independent on the macroeconomic environment. However, the contribution of technology transfers is more sensitive during financially constrained periods for innovations linked to patents.

6.1.3 Differentiating by Pavitt sectoral taxonomy

In this sub-section, we deepen our analysis by looking at whether there are differences in the effects of technology transfers from universities on firm innovativeness across sectors that might have different underlying technological appropriability and patterns of innovation. For this purpose, we follow the sectoral taxonomy purpose by Pavitt (1984) that has been broadly used in the innovation literature (Dosi, 1988). Pavitt (1984) distinguishes between four different major industries: science-based, specialized-supplier, scale-intensive and supplier-dominated sectors.

In Table 9, we present the ATT estimates using the CDiD methodology differentiating between the four sectoral groups. In panel A, we show results for *science-based* sectors. These sectors are extremely innovative and highly R&D intensive (for example, electronics, pharmaceutical or chemical industries are in this group). In panel B, we present the estimations for *specialized-supplier* sectors. These industries are typically characterized by small firms with strong links to their users. Firms operating in these sectors have opportunities to innovate although formal R&D expenditures tend to be small. For example, manufacturers of electrical machinery or mechanical engineering are included in this group. In panel C, we show estimates for *scale-intensive* sectors. These industries tend to profit from economies of scale and therefore firms in these sectors have strong incentives to innovate. Examples of these sectors are metal manufacturing or transport equipment. In panel D, we present the results for *supplier-dominated* sectors. These are traditional sectors such as agriculture or textiles where incremental innovations are more common than drastic innovations and, therefore, appropriability tends to be small. Balancing tests for the different sectors are presented in Table OA11 in the on-line Appendix.

The results suggest that the effects of technology transfers from universities are particularly important for science-based and scale and information intensive sectors. In these two sets of sectors, technology transfers from universities lead to an increase in product innovation by 8.4 percentage points and by 18.2 percentage points, respectively. The results from panel A for science-based sectors also show a positive and statistically significant effect of technology transfers on patents. For the rest of the panels and innovative indicators the estimated coefficients are positive although not significant at standard statistical levels, with the exception of panel D, column (1a) for supplier dominated sectors and product innovation, where the estimated coefficient is negative although not significantly different from

zero. These findings suggest that sectors that are highly R&D intensive and with high levels of appropriability benefit relatively more from technology transfers from universities than less R&D intensive sectors. However, we believe that these results need to be taken cautiously because the stratification of the sample leads to a small number of firms in each group. Therefore, this evidence is only suggestive of the potential differential effects of university technology transfers by innovative sectors.

6.2. Contribution of technology transfers beyond the direct effect on innovation

6.2.1. Spillover effects

Our identification assumption for calculating the effects of technology transfers on innovation is that technology transfers do not generate spillovers on the control group (Stable Unit Treatment Value Assumption, SUTVA). To investigate whether there is a bias in our previous estimations, and if there is a bias in its direction, we study spillover or indirect effects. Our underlying assumption is that spillovers are regional- and industry-concentrated (Griliches, 1992; Jaffe et al., 1993; Agrawal et al., 2017). Our measure of spillovers is calculated in the spirit of Girma et al. (2015) or García-Vega et al. (2019). We measure the difference in innovation output of firms without technology transfers from universities in clusters where there is a high concentration of firms with technology transfers from universities (treated group) and firms without technology transfers from universities in clusters with low concentration of firms with technology transfers with an average of 5.4% of firms with technology transfers from universities. We consider clusters with high concentration of technology transfers as those clusters with technology transfers above the median and we run robustness checks with thresholds at the 80th and 90th percentile of the distribution of technology transfers from universities.

²¹ The results obtained with thresholds at the 80th and 90th percentile of technology transfers from universities (not reported) are similar to those presented in Table 10.

We present the estimated ATT effects using the CDiD methodology in Table 10 and the balancing test in Table OA12 in the on-line Appendix. We do not find any statistically significant effect for process innovation in column 2. However, we find a positive and statistically significant effect for product innovation and patents, in columns 1 and 3 respectively. This result suggests that there are positive spillovers, but they are only statistically significant for product innovation and patents. Therefore, technology transfers from universities seem to have an important contribution to firm innovation in addition to the uncovered direct effects: Firms that do not acquire technology from universities also profit from technology of universities in order to patent or to improve their products if they are located in regions and in industries with high concentrations of contractual technology transfers. Since the spillovers are positive, they imply that our estimates of the direct effects for patents are a lower bound of the technology transfers from universities' effect on firm innovation.

6.2.2. Crowding-out effects

We next turn to the question of whether technology transfers from universities is crowding-out the internal R&D of the firm. The study of complementarities or substitutability between technology sourcing and internal R&D has long been of interest to the literature on R&D governance (Barge-Gil et al., 2018, Mohnen and Röller, 2005, among others). For example, Cassiman and Veugelers (2006), using cross-sectional data on Belgian firms, find that external R&D is a complement to the R&D conducted in-house in order to generate innovations. A related paper is Ceccagnoli et al. (2014), who study the sources of complementarity between internal and external R&D. Examining a sample of pharmaceutical companies, these authors find that internal and external R&D are largely independent and that complementarity depends on a buyer's characteristics, such as absorptive capacity, economies of scale and external R&D are complementary innovation strategies for increasing total factor productivity. In our approach, we do not formally perform a test for complementarity or substitutability in order to generate innovations, which is beyond the scope of this paper; instead, we study whether technology transfers from universities lead to an increase in firm innovation inputs. The logic is that if firms reduce their innovation inputs after having technology transfers from universities, it would indicate that firms

are substituting internal knowledge with external knowledge from universities. In the long run, this could damage the internal capabilities of the firms.

For our analysis, we consider as innovation input two different types of R&D expenditures and the number of researchers working in R&D. We present evidence of the effect of technology transfers from universities on R&D inputs using the CDiD methodology in Table 11. In column 1, we analyse the logarithm of total innovation expenditures (this includes internal R&D expenditures or intra-mural R&D and other expenditures such as training for workers, product alternations, market research and advertising); in column 2, we study the logarithm of internal R&D; and in column 3, the input variable is the logarithm of the number of researchers working in R&D in the firm.

In all cases, technology transfers from universities has a positive and statistically significant effect. The estimates show that having technology transfers from universities increases total innovation expenditures by 32.1%, internal R&D expenditures by 18.3% and researchers in R&D by 25.6%. These results suggest that there are no crowding-out effects and that technology transfers from universities lead to an increase in firm innovation inputs and job creation in high-skill jobs.

7. Summary and concluding remarks

To gain a better understanding of the contribution that university knowledge makes to private firms, and thus indirectly to society, this paper studies the effect of technology transfers from universities on firm innovativeness. We find that technology transfers from universities have an important positive effect on firm innovativeness. We also show that this effect holds especially during the non-recession period. Moreover, our results suggest that technology transfers induce positive spillovers and increase the internal capabilities of firms.

These results are consistent with universities providing superior technologies and, thus, allowing firms to profit from types of knowledge that cannot be easily obtained internally. We show that this frontier knowledge benefits particularly small firms. Thus, our results imply that universities play a

significant role in the innovation of small firms and, consequently universities are important in an indirect way for local job creation. Typical of SMEs, liquidity constraints and difficulties on attracting highly skilled workers are barriers hampering their innovation performance. By facilitating access to specialized expert knowledge through university technology transfers, universities help to overcome these barriers and improve firm competitiveness. Furthermore, given the additionality that we find of technology transfers from universities on in-house R&D, the further promotion of university technology transfers might enhance the absorptive capacity of firms and, hence, their productivity.

The decrease of the strength of university technology transfers during the crisis period may be a consequence of the important pay cuts suffered by Spanish universities, which affected the quality of the knowledge transferred. Our results suggest that the public sector should try to maintain its support to universities, also in times of recession. Finally, the spillover effects of technology transfers found in this study make the above recommendations even more pertinent. In other words, financing universities has a private benefit, but also a benefit for the economy as a whole, through the upgrade on firm innovativeness operating in the same region and sector.

Although this study provides relevant insights, we acknowledge some limitations. First, the results are obtained based on data from a single country. Although the theoretical arguments suggests that our empirical results can be generalized to other countries, it would be interesting to extend the analysis to other countries. For example, it is possible that the impact of technology transfer on firm innovativeness varies with the type of ownership of the university, and therefore, in countries with a larger proportion of private universities than Spain, the estimated impact might be larger. Second, we do not have information about informal contacts between firms and universities, which are also part of knowledge transfers. This could lead to a downward bias, which would mean that our results are a very conservative measure of technology transfers. Moreover, we do not have information about the type of knowledge firms are getting from universities, either. Hence, we are not able to disentangle the different effects upon firms' innovativeness depending on the type of knowledge transferred. Finally, it is possible that the experience of providing knowledge to firms also provides a learning opportunity for university researchers. These are questions that are interesting avenues for future research.

The evidence we present is relevant for the understanding of the social value of universities beyond education. Our study highlights the importance of universities to contribute to the innovations in the private sector in a direct and indirect way. Hence, our results support the idea that knowledge creation by universities provides an important public good to society. Overall, our findings suggest that when policy makers are looking for policies to promote innovation and local growth, they should also consider public funding of universities.

References

- Aerts, K. and T. Schmidt (2008). Two for the price of one?: Additionality effects of R&D subsidies: A comparison between Flanders and Germany. *Research Policy*, 37(5): 806-822.
- Aghion, P., Akcigit, U., and P. Howitt (2015). The Schumpeterian Growth Paradigm. *Annual Review* of Economics, 7: 557-575.
- Aghion, P., Askenazy, P., Berman, N., Cette, G. and L. Eymard (2012). Credit constraints and the cyclicality of R&D investment: Evidence from France. *Journal of the European Economic Association*, 10(5): 1001-1024.
- Aghion, P., Blundell, R., Griffith, R., Howitt, P., and S. Prantl (2009). The effects of entry on incumbent innovation and productivity. *The Review of Economics and Statistics*, 91(1): 20-32.
- Aghion, P., Van Reenen, J., and L. Zingales (2013). Innovation and institutional ownership. *American Economic Review*, 103(1): 277-304.
- Agrawal, A., Galasso, A. and A. Oettl (2017). Roads and innovation. *The Review of Economics and Statistics*, 99(3): 417-434.
- Almeida, P., Dokko, G., and L. Rosenkopf (2003). Startup size and the mechanisms of external learning: increasing opportunity and decreasing ability? *Research Policy*, *32*(2), 301-315.
- Almunia, M., Antràs, P., Lopez-Rodriguez, D., and E. Morales (2018). Venting Out: Exports during a domestic slump. Banco de España Working Paper No. 1844. Available at SSRN: <u>https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3303155</u>.
- Antonelli, C. (1999). The evolution of the industrial organisation of the production of knowledge. *Cambridge journal of economics*, 23(2), 243-260.
- Añón Higón, D., Máñez, J.A. and J.A. Sanchis-Llopis (2018). Intramural and external R&D: evidence for complementarity or substitutability. *Economia Politica*, 35(2): 555–577.
- Appleyard, M (1996). How does knowledge flow? Interfirm patterns in the semiconductor industry. *Strategic management journal*, 17(S2): 137-154.
- Arvanitis, S., Kubli, U., and M. Woerter (2008). University-industry knowledge and technology transfer in Switzerland: What university scientists think about co-operation with private enterprises. *Research Policy*, 37(10): 1865-1883.
- Audretsch, D., Segarra, A., and M. Teruel (2014). Why don't all young firms invest in R&D?. *Small Business Economics*, 43(4), 751-766.
- Autio, E., Kenney, M., Mustar, P., Siegel, D., and M. Wright (2014). Entrepreneurial Innovation: The Importance of the Context. *Research Policy*, 43: 1097-1108.
- Aw, B. Y., M. J. Roberts and D. Y. Xu (2011). R&D investment, exporting and productivity dynamics, *American Economic Review*, 101, 1312–1344.
- Azoulay, P., Graff Zivin, J. S., Li, D., and B. Sampat (2018). Public R&D investments and privatesector patenting: evidence from NIH funding rules. *The Review of Economic Studies*, 86(1): 117-152.

- Barge-Gil, A., Huergo, E., López, A. and L. Moreno (2018). Empirical models of firms' R&D, in L. C. Corchón and M. A. Marini (ed.): *Handbook of Game Theory and Industrial Organization*, Edward Elgar. Cheltenham, UK and Northampton, MA.
- Baumol, W.J., 2002. Entrepreneurship, innovation and growth: The David-Goliath symbiosis. *Journal* of Entrepreneurial Finance 7(2): 1-10.
- Beise, M., and H. Stahl (1999). Public research and industrial innovations in Germany. *Research policy*, 28(4): 397-422.
- Belenzon, S., and M. Schankerman (2009). University knowledge transfer: private ownership, incentives, and local development objectives. *The Journal of Law and Economics*, 52(1): 111-144.
- Bena, J., and K. Li (2014). Corporate innovations and mergers and acquisitions. *The Journal of Finance*, 69(5): 1923-1960.
- Beneito, P., Rochina-Barrachina, M.E., and A. Sanchis-Llopis (2015). Ownership and the cyclicality of firms' R&D investment. *International entrepreneurship and management Journal*, 11(2): 343-359.
- Bercovitz, J., and M. Feldman (2007). Fishing upstream: Firm innovation strategy and university research alliances. *Research policy*, 36(7): 930-948.
- Bikard, M. (2018). Made in academia: The effect of institutional origin on inventors' attention to science. *Organization Science*, 29(5), 818-836.
- Bishop, K., D'Este, P., and A. Neely (2011). Gaining from interactions with universities: Multiple methods for nurturing absorptive capacity. *Research Policy*, 40(1): 30-40.
- Bloom, N., Jones, C. Van Reenen J., and M. Webb (2017). Are ideas getting harder to find?. NBER Working Paper No. w23782
- Blundell, R. and M.C. Dias (2002). Alternative approaches to evaluation in empirical microeconomics. *Portuguese economic journal*, 1(2): 91-115.
- Caldera, A., and O. Debande (2010). Performance of Spanish universities in technology transfer: An empirical analysis. *Research policy*, 39(9): 1160-1173.
- Cassiman, B. and R. Veugelers (2006). In Search of Complementarity in Innovation Strategy: Internal R&D and External Technology Acquisition. *Management Science*, 52(1): 68–82.
- Cassiman, B., Di Guardo, M. C., and G. Valentini (2010). Organizing links with science: Cooperate or contract?: A project-level analysis. *Research Policy*, *39*(7), 882-892.
- Ceccagnoli, M., Higgins, M. and V. Palermo (2014). Behind the scenes: Sources of complementarity in R&D. *Journal of Economics & Management Strategy*, 23 (1): 125-148.
- Cefis, E., and L. Orsenigo (2001). The persistence of innovative activities: a cross-countries and cross-sectors comparative analysis. *Research Policy*, 30(7): 1139–1158.
- Cefis, E. (2003). Is there persistence in innovative activities?. *International Journal of Industrial Organization*, 21: 489–515.
- Chapple, W., Lockett, A., Siegel, D., and M. Wright (2005). Assessing the relative performance of UK university technology transfer offices: parametric and non-parametric evidence. *Research Policy*, 34(3): 369-384.

- Clausen, T., Pohjola, M., Sapprasert, K., and B. Verspagen (2011). Innovation strategies as a source of persistent innovation. *Industry and Corporate Change*, 21 (3): 553–585.
- Cohen, W. M., Nelson, R. R., and J. Walsh (2002). Links and impacts: The influence of public research on industrial R&D. *Management Science*, 48(1): 1–23.
- Cosh, A., Hughes, A., and R. Lester (2006). *UK plc: Just How Innovative Are We?* Cambridge, UK, Cambridge–MIT Institute.
- Dasgupta, P., and P. David (1994). Towards a new economics of science. *Research Policy*, 23(5), 487–521.
- Decker, R., Haltiwanger, J., Jarmin, R., and J. Miranda (2014). The Role of Entrepreneurship in US Job Creation and Economic Dynamism. *Journal of Economic Perspectives*, 28(3): 3-24.
- D'Este, P., and P. Patel (2007). University–industry linkages in the UK: What are the factors underlying the variety of interactions with industry?. *Research policy*, 36(9): 1295-1313.
- Dosi, G. (1988). Sources, procedures, and microeconomic effects of innovation. *Journal of economic literature* 1120-1171.
- Fudickar, R., and H. Hottenrott (2019). Public research and the innovation performance of new technology based firms. *The Journal of Technology Transfer*, 44(2): 326-358.
- Ganter, A., and A. Hecker (2013). Deciphering antecedents of organizational innovation. *Journal of Business Research*, 66(5): 575-584.
- García-Vega, M., P. Hofmann and R. Kneller (2019). The internationalisation of R&D and the knowledge production function. *International Journal of Industrial Organization*, 63: 583-614.
- Geroski, P., Van Renen, J., and C. Walters (1997). How persistently do firms innovate?. *Research Policy*, 26: 33–48.
- Girma, S., Gong, Y., Görg, H. and S. Lancheros (2015). Estimating Direct and Indirect Effects of Foreign Direct Investment on Firm Productivity in the Presence of Interactions between Firms. *Journal of International Economics*, 95(1): 157-169.
- Griffith, R., Huergo, E., Mairesse, J., and B. Peters (2006). Innovation and productivity across four European countries. *Oxford review of economic policy*, 22(4): 483-498.
- Griliches, Z. (1992). The Search for R&D spillovers. Scandinavian Journal of Economics, 94: 29-47.
- Grimaldi, R., and N. Von Tunzelmann (2002). Assessing collaborative, pre-competitive R&D projects: the case of the UK LINK scheme. *R&D Management*, *32*(2), 165-173.
- Guadalupe, M., Kuzmina, O., and C. Thomas (2012). Innovation and foreign ownership. *American Economic Review*, *102*(7): 3594-3627.
- Haltiwanger, J., Jarmin, R., and J. Miranda (2013). Who Creates Jobs?: Small versus Large versus Young. *The Review of Economics and Statistics*, Vol. XCV, No. 2: 347-361.
- Haucap, J., Rasch, A., and J. Stiebale (2019). How mergers affect innovation: Theory and evidence. *International Journal of Industrial Organization*, 63: 283-325.

- Jabbour, L., Tao, Z., Vanino, E., and Y. Zhang (2019). The good, the bad and the ugly: Chinese imports, European Union anti-dumping measures and firm performance. *Journal of International Economics*, 117: 1-20.
- Jaffe, A. B., M. Trajtenberg, and R. Henderson (1993). Geographical Localization of Knowledge Spillovers as Evidenced by Patent Citations. *Quarterly Journal of Economics*, 108: 577-98.
- Javorcik, B., and S. Poelhekke (2017). Former foreign affiliates: Cast out and outperformed?. *Journal of the European Economic Association*, 15(3): 501-539.
- Kafouros, M., Wang, C., Piperopoulos, P., and M. Zhang (2015). Academic collaborations and firm innovation performance in China: The role of region-specific institutions. *Research Policy*, 44(3), 803-817.
- Lach, S., Neaman, Z., and M. Schankerman (2017). Government Financing of R&D: A Mechanism Design Approach. CEPR Discussion Paper 12199.
- Lane, P.J., and M. Lubatkin, (1998). Relative absorptive capacity and interorganizational learning. *Strategic management journal*, 19(5), 461-477.
- López, A. (2008). Determinants of R&D cooperation: Evidence from Spanish manufacturing firms. *International Journal of Industrial Organization*, 26(1), 113-136.
- López-García, P., Montero, J. M., and E. Moral-Benito (2013). Business cycles and investment in productivity-enhancing activities: evidence from Spanish firms. *Industry and Innovation*, 20(7): 611-636.
- Lööf, H., & Broström, A. (2008). Does knowledge diffusion between university and industry increase innovativeness?. *The Journal of Technology Transfer*, 33(1), 73-90.
- Lucena, A. (2011). The organizational designs of R&D activities and their performance implications: empirical evidence for Spain. *Industry and innovation*, *18*(02), 151-176.
- Macho-Stadler, I., Pérez-Castrillo, D., and R. Veugelers (2007). Licensing of university inventions: The role of a technology transfer office. *International Journal of Industrial Organization*, 25(3): 483-510.
- Mansfield, E. (1991). Academic research and industrial innovation. *Research policy*, 20(1): 1-12.
- Mairesse, J. and P. Mohnen (2005). The importance of R&D for innovation: A reassessment using French survey data. *Journal of Technology Transfer*, 30(1-2): 183-197.
- Martínez-Ros, E., and J. Labeaga (2009). Product and process innovation: Persistence and complementarities. *European Management Review*, 6(1): 64-75.
- McGuirk, H., Lenihan, H. and M. Hart (2015). Measuring the impact of innovative human capital on small firms' propensity to innovate. *Research Policy*, 44(4): 965-976.
- Medda, G., Piga, C. and D. Siegel (2005). University R&D and firm productivity: evidence from Italy. *The Journal of Technology Transfer*, 30(1-2): 199-205.
- Miotti, L., and F. Sachwald (2003). Co-operative R&D: why and with whom?: An integrated framework of analysis. *Research policy*, *32*(8), 1481-1499.
- Mohnen, P. and L. Röller (2005). Complementarities in innovation policy. *European Economic Review*, 49(6): 1431-1450.

- Monjon, S., and P. Waelbroeck (2003). Assessing spillovers from universities to firms: evidence from French firm-level data. *International Journal of Industrial Organization*, 21(9), 1255-1270.
- Monteiro, F. and J. Birkinshaw (2017). The external knowledge sourcing process in multinational corporations. *Strategic Management Journal*, 38(2): 342-362.
- Negassi, S. (2004). R&D co-operation and innovation a microeconometric study on French firms. *Research policy*, 33(3), 365-384.
- Novartis, (2017). Annual report, available at: <u>https://www.novartis.com/sites/www.novartis.com/files/novartis-annual-report-2017-en.pdf</u>.
- OECD (2005a). Oslo Manual. Guidelines for Collecting and Interpreting Innovation Data. *OECD Publishing*, Paris.
- OECD (2005b). OECD SME and Entrepreneurship Outlook: 2005, OECD Paris.
- OECD (2018). Strengthening SMEs and entrepreneurship for productivity and inclusive growth. Available at: <u>https://www.oecd.org/cfe/smes/ministerial/documents/2018-SME-Ministerial-Conference-Key-Issues.pdf</u>.
- Parmigiani, A. (2007). Why do firms both make and buy? An investigation of concurrent sourcing. *Strategic Management Journal*, 28(3): 285-311.
- Pavitt, K. (1984). Sectoral patterns of technical change: towards a taxonomy and a theory. *Research policy*, 13(6): 343-373.
- Perkmann, M., and K. Walsh (2007). University-industry relationships and open innovation: Towards a research agenda. *International Journal of Management Reviews*, 9(4): 259-280.
- Piga, C. and M. Vivarelli (2004). Internal and external R&D: a sample selection approach. *Oxford Bulletin of Economics and Statistics*, 66(4): 457-482.
- Rosenbaum, P.R. and D.B. Rubin (1983). The central role of the propensity score in observational studies for causal effects. *Biometrika* 70 (1): 41–55.
- Rosenbaum, P.R., and D.B. Rubin (1985). Constructing a control group using multivariate matched sampling methods that incorporate the propensity score. *Journal of the American Statistical Association* 39 (1): 33–38.
- Rosenberg, N. and R. Nelson (1994). American universities and technical advance in industry. *Research Policy*, 23: 323–348.
- Russell Group (2010). *The Economic Impact of Research Conducted in Russell Group Universities*. Russell Group Papers, 1/2010.
- Sacristán, V. (2017). ¿Quién financia la universidad? Comparación entre comunidades autónomas en España, Europa y la OCDE, 2009-2015. OSU: Observatorio Sistema Universitario.
- Scherer, F. M., and D. Harhoff (2000). Technology Policy for a World of Skew-distributed Outcomes. *Research Policy*, 29(4–5): 559–66.
- Seru, A. (2014). Firm boundaries matter: Evidence from conglomerates and R&D activity. *Journal of Financial Economics*, 111(2): 381-405.

- Siegel, D. S., Waldman, D., and A. Link (2003a). Assessing the impact of organizational practices on the relative productivity of university technology transfer offices: an exploratory study. *Research policy*, 32(1): 27-48.
- Siegel, D., Waldman, D., Atwater, L., and A. Link (2003b). Commercial knowledge transfers from universities to firms: improving the effectiveness of university–industry collaboration. *The Journal of High Technology Management Research*, 14(1): 111-133.
- Siegel, D., Waldman, D., Atwater, L. and A. Link (2004). Toward a model of the effective transfer of scientific knowledge from academicians to practitioners: qualitative evidence from the commercialization of university technologies. *Journal of engineering and technology management*, 21(1-2): 115-142.
- Siegel, D. S., Veugelers, R., and M. Wright (2007). Technology transfer offices and commercialization of university intellectual property: performance and policy implications. *Oxford review of economic policy*, 23(4): 640-660.
- Siemer, M. (2019). Employment effects of financial constraints during the Great Recession. *Review of Economics and Statistics*, 101(1): 16-29.
- Stephan, P. (1996). The Economics of Science. Journal of Economic Literature, 34, 1199-1235.
- Stiebale, J. (2016). Cross-border M&As and innovative activity of acquiring and target firms. *Journal of International Economics*, 99: 1-15.
- Stiebale, J. and N. Woessner (2019). M&As, investment and financing constraints. *International Journal of the Economics of Business*, Sept-5: 1-44.
- Stock, J. H. and M. Yogo (2002). Testing for Weak Instruments in Linear IV Regression, NBER Technical Working Paper no 284.
- Stuart, E. (2010). Matching methods for causal inference: A review and a look forward, *Statistical Science* Feb 1; 25(1): 1–21.
- Szücs, F. (2018). Research subsidies, industry–university cooperation and innovation. *Research Policy*, 47(7), 1256-1266.
- Tavassoli, S., and C. Karlsson (2015). Persistence of various types of innovation analyzed and explained. *Research Policy*, 44(10): 1887-1901.
- Teirlinck, P., and A. Spithoven (2013). Research collaboration and R&D outsourcing: Different R&D personnel requirements in SMEs. *Technovation*, *33*(4-5): 142-153.
- Thomas, N.R. (2010). Frederic Stanley Kipping—Pioneer in Silicon Chemistry: His Life & Legacy. *Silicon*, 2(4):187-193.
- Vega-Jurado, J., Kask, S., and L. Manjarrés-Henriquez (2017). University industry links and product innovation: cooperate or contract? *Journal of Technology Management & Innovation*, 12(3): 1-8.
- Veugelers, R. (2016). The embodiment of knowledge: universities as engines of growth. Oxford Review of Economic Policy, 32(4): 615-631.
- Williamson, O. E. (1985). The Economic Institutions of Capitalism: Firms, Markets, Relational Contracting. New York: Free Press.

- Yu, G. J., and J. Lee (2017). When should a firm collaborate with research organizations for innovation performance? The moderating role of innovation orientation, size, and age. *The Journal of Technology Transfer*, 42(6), 1451-1465.
- Wooldridge, J. M. (2005). Simple solutions to the initial conditions problem in dynamic, nonlinear panel data models with unobserved heterogeneity. *Journal of applied econometrics*, 20(1): 39-54.

TABLES

Years	Total	With techno. transfers from universities	First time starters	Never with transfers	Without techno. transfers but with overlap
	(1)	(2)	(3)	(4)	(5)
2005	12,098	909	457	9,806	1,383
2006	12,034	935	288	9,735	1,364
2007	11,594	915	229	9,336	1,343
2008	11,182	805	140	8,968	1,409
2009	10,796	702	106	8,632	1,462
2010	10,380	686	99	8,259	1,435
2011	9,977	597	81	7,915	1,465
2012	9,612	526	64	7,607	1,479
2013	9,172	464	67	7,240	1,468

Table 1: Number of firms

Note: *Total* is the total number of firms in each year in the sample. *With techno. transfers from universities* is the number of firms in a given year with positive expenditures in R&D services from Spanish Universities. *First time starters* is the number of firms in a given year that start for the first time to obtain technology transfers from Universities. *Never with transfers* is the number of firms in a given year that do not have technology transfers from Universities in that year and that they never had it in the past and never will have it in the future during the sample period. *Without techno. transfers but with overlap* is the number of firms in a given year without technology transfers from Universities but with either technology transfers in the past or with technology transfers in the period.

Table 2: Descriptive statistics distinguishing between full sample, firms with technology transfers from universities and without technology transfers
from universities

		Full sam	ple	Wit	h technology	y transfers	Withou	ut technology	transfers
Variable	Mean	Std. Dev.	Observations	Mean	Std. Dev.	Observations	Mean	Std. Dev.	Observations
Treatment variable									
Technology transfers from universities	0.068	0.251	114009	1.000	0.000	7714	0.000	0.000	106295
Outcome variables									
Product innovation	0.471	0.499	114009	0.749	0.433	7714	0.451	0.498	106295
Process innovation	0.485	0.500	114009	0.717	0.451	7714	0.468	0.499	106295
Patents	0.103	0.303	114009	0.309	0.462	7714	0.088	0.283	106295
Sales from products new to the market	3.764	6.308	114009	7.498	7.390	7714	3.493	6.135	106295
Sales from products new to the firm	5.011	6.867	114009	8.027	7.420	7714	4.792	6.774	106295
Number of patents	0.123	0.440	103916	0.433	0.808	6945	0.101	0.392	96971
Control variables									
Employment	4.122	1.720	114004	4.426	1.659	7714	4.100	1.723	106290
Physical investment	7.824	2.456	79069	8.449	2.501	6983	7.764	2.444	72086
Internal R&D expenditures	7.751	1.573	55292	8.615	1.595	7232	7.621	1.528	48060
Innovation expenditures	7.845	1.759	66848	8.901	1.677	7714	7.707	1.722	59134
Sales	11.282	2.141	113907	11.798	2.324	7707	11.244	2.123	106200
Exports	0.593	0.491	114009	0.754	0.430	7714	0.581	0.493	106295
Group	0.400	0.490	114009	0.496	0.500	7714	0.393	0.488	106295

Note: Technology transfers from universities is a dummy variable that takes the value one if a firm has expenditures in R&D services from Spanish Universities. Product (process) innovation is a dummy variable that takes the value one if a firm reports having introduced new or significantly improved products (production processes) in the current or previous two years. Patents is a dummy variable that takes the value one if a firm reports having patents in the current or previous two years. Sales from products new to the market (firm) is the natural logarithm of the sales that come from new-to-the-market (new-to-the-firm) products in a current year. Number of patents is the natural logarithm of the number of employees. Physical investment is the natural logarithm of the physical investments of the firm. Internal R&D expenditures is the natural logarithm of the sales of the company. Exports is a dummy variable that takes the value one if the firm exports. Group is a dummy variable that takes the value one if the firm exports. Group is a dummy variable that takes the value one if the firm belongs to a business group.

Table 3: The effect of technology transfers from universities on firms' innovation: Effects on the	
unmatched sample	

Panel A: Dependent variable pro-	duct innova	ition			
	(1a)	(2a)	(3a)	(4a)	(5a)
University technology transfers	0.242***	0.184***	0.074***	0.028***	0.028***
	(0.009)	(0.010)	(0.008)	(0.008)	(0.008)
Observations	83,854	58,395	50,675	58,395	58,395
R-squared	0.120	0.117		0.045	0.049
Number of id			10,806	11,329	11,329
Panel B: Dependent variable pro	cess innova	tion			
	(1b)	(2b)	(3b)	(4b)	(5b)
University technology transfers	0.204***	0.131***	0.052***	0.023***	0.023***
	(0.010)	(0.010)	(0.008)	(0.009)	(0.009)
Observations	83,854	58,395	50,675	58,395	58,395
R-squared	0.068	0.075		0.051	0.055
Number of id			10,806	11,329	11,329
Panel C: Dependent variable pat	ents				
	(1c)	(2c)	(3c)	(4c)	(5c)
University technology transfers	0.180***	0.160***	0.035***	0.020***	0.019***
	(0.010)	(0.010)	(0.004)	(0.007)	(0.007)
Observations	83,854	58,395	50,339	58,395	58,395
R-squared	0.064	0.078	-	0.005	0.008
Number of id			10,795	11,329	11,329
Sector FEs	Yes	Yes		Yes	
Firm FEs				Yes	Yes
Lagged control variables		Yes	Yes	Yes	Yes
Firm FEs Wooldridge correction			Yes		
Sector x time FEs			Yes		Yes
Year FEs in all regressions					

Notes: In all columns, we estimate linear probability models with the exception of column (3), where we estimate a probit model. University technology transfers is a dummy variable that takes the value one if a firm has expenditures in R&D services from Spanish universities. The controls are the lagged values of the following variables: the natural logarithm of the number of employees, the natural logarithm of the physical investments. For exact definitions and sources of all variables see Tables1 and 2 and the main text. Estimated robust standard errors clustered at the firm level are in parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Dependent variable	Product innovation	Process innovation	Patents	Product innovation	Process innovation	Patents
*	(1)	(2)	(3)	(4)	(5)	(6)
University	0.078***	0.069***	0.093***			
technology transfers	(0.023)	(0.025)	(0.020)			
Quartiles of technolog	y transfer					
$\tilde{1}^{\text{st}}$ quartile				0.126***	0.145***	0.122***
*				(0.030)	(0.034)	(0.034)
2 nd quartile				0.089**	0.084**	0.127***
				(0.036)	(0.041)	(0.037)
3 rd quartile				0.065*	0.032	0.078**
				(0.033)	(0.038)	(0.031)
4 th quartile				0.021	0.009	0.042
				(0.040)	(0.041)	(0.032)
Observations	1,416	1,416	1,416	1,416	1,416	1,416
Number of id	1,360	1,410	1,410	1,410	1,360	1,410

Table 4: Effect of technology transfers from universities on firms' innovation. Average treatment effect on the treated after matching for starters

Notes: University technology transfers is a dummy variable that takes the value one if a firm has expenditures in R&D services from Spanish Universities. The treated group are starters (firms that for the first time in our sample period have technology transfers from universities). The control group are firms that never have technology transfers from universities in our sample period. Quartiles of technology transfers are the different quartiles of university technology transfers intensity, where intensity is the ratio between R&D acquisitions and total R&D expenditures. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table 5: Effect of technology transfers from universities on firms' innovation. Average treatment effect on the treated after matching with CDiD methodology

	Product		
Dependent variable	innovation	Process innovation	Patents
_	(1)	(2)	(3)
University technology transfers ^{CDiD}	0.043*	0.072**	0.055**
	(0.025)	(0.027)	(0.021)
Observations	5,841	5,841	5,841
Number of id	3,462	3,462	3,462

Notes: University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table 6: Technology transfers from universities on firms' innovation with respect to technology transfers from other providers. Average treatment effect on the treated after matching with CDiD methodology

	Product		
Dependent variable	innovation	Process innovation	Patents
_	(1)	(2)	(3)
University technology transfers ^{CDiD}	0.136**	0.137**	0.125*
	(0.055)	(0.063)	(0.064)
Observations	1,157	1,157	1,157
Number of id	807	807	807

Treated group: Companies with technology transfers from universities *Control group*: Companies with R&D acquisitions from private companies and other institutions that are not universities

Notes: University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table 7: Effect of technology transfers from universities for SMEs vs non-SMEs. Average treatment effect on the treated after matching with CDiD methodology

		SMEs			Non-SMEs		
Dependent variable	Product innovation	Process innovation	Patents	Product innovation	Process innovation	Patents	
-	(1)	(2)	(3)	(4)	(5)	(6)	
University	0.026	0.063*	0.098***	0.042	0.044	0.090	
technology transfers ^{CDiD}	0.030	0.032	0.023	0.071	0.075	0.083	
Observations	4,208	4,208	4,208	574	574	574	
Number of id	2,668	2,668	2,668	413	413	413	

Notes: SMEs are firms with at most 250 employees. University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables A1 and A2 in the Appendix. Standard errors between parentheses. * Significant at 10%;** Significant at 5%; *** significant at 1%.

Table 8: Effect of technology transfers from universities during the recession and non-recession period. Average treatment effect on the treated after matching with CDiD methodology

	Non- recession period			Re	Recession period		
	Product	Process	Patents	Product	Process	Patents	
Dependent variable	innovation	innovation		innovation	innovation		
	(1)	(2)	(3)	(4)	(5)	(6)	
University	0.062	0.077*	0.133***	0.026	0.098**	0.016	
technology transfers ^{CDiD}	(0.040)	(0.044)	(0.034)	(0.038)	(0.041)	(0.031)	
Observations	1,968	1,968	1,968	2,645	2,645	2,645	
Number of id	1,775	1,775	1,775	1,978	1,978	1,978	

Notes: University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Dependent vorichle	Draduct innervation	Drococc innervation	Dotonto
Dependent variable	Product innovation	Process innovation	Patents
Panel A: Science based	(1a)	(2a)	(3a)
University technology transfers ^{CDiD}	0.084*	0.046	0.075*
	(0.051)	(0.059)	(0.042)
Observations	1,217	1,217	1,217
Number of id	821	821	821
Panel B: Specialized suppliers	(1b)	(2b)	(3b)
University technology transfers ^{CDiD}	0.066	0.147	0.132
	(0.073)	(0.091)	(0.083)
Observations	570	570	570
Number of id	374	374	374
Panel C: Scale and information intensive	(1c)	(2c)	(3c)
University technology transfers ^{CDiD}	0.182***	0.007	0.009
	(0.059)	(0.058)	(0.053)
Observations	1,133	1,133	1,133
Number of id	689	689	689
Panel D: Supplier dominated	(1d)	(2d)	(3d)
University technology transfers ^{CDiD}	-0.090	0.067	0.061
	(0.062)	(0.060)	(0.043)
Observations	1,067	1,067	1,067
Number of id	644	644	644

 Table 9: Effect of technology transfers from universities differentiating by sectoral Pavitt taxonomy. Average treatment effect on the treated after matching with CDiD methodology

Notes: University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table 10: Spillover effect. Average treatment effect on the treated after matching with CDiD methodology

Dependent variable	Product innovation	Process innovation	Patents
	(1)	(2)	(3)
Spillover	0.076*	-0.024	0.095**
-	(0.042)	(0.048)	(0.038)
Observations	1,460	1,460	1,460
Number of id	1,008	1,008	1,008

Treated group: Companies without technology transfers from universities located in regions and sectors with **high** technology transfers from universities.

Control group: Companies without technology transfers from universities located in regions and sectors with **low** technology transfers from universities

Notes: For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table 11: Effect of technology transfers from universities on firms' internal R&D capabilities. Average treatment effect on the treated after matching with CDiD methodology

Dependent variable:	Total R&D expenditures	Internal R&D expenditures	Researchers
	(1)	(2)	(3)
University technology transfers ^{CDiD}	0.321***	0.183**	0.256***
	(0.097)	(0.091)	(0.070)
Observations	5,841	5,841	5,841
Number of id	3,462	3,462	3,462

Notes: University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Appendix

Propensity score for:	One-to-one matching	CDiD methodology
	(1)	(2)
Product innovation _{t-1}	-0.002	-0.002
	(0.002)	(0.004)
Process innovation _{t-1}	0.003	0.018***
	(0.002)	(0.003)
Patents _{t-1}	0.007**	0.050***
	(0.003)	(0.004)
Employment _{t-1}	-0.001	-0.005***
	(0.001)	(0.002)
Physical capital _{t-1}	0.002***	0.003***
	(0.001)	(0.001)
Researchers in R&D _{t-1}	0.003***	0.014***
	(0.001)	(0.002)
Internal R&D _{t-1}	0.010***	0.039***
	(0.001)	(0.001)
Exports _{t-1}	0.001	0.012***
	(0.002)	(0.004)
Group _{t-1}	-0.001	-0.004
	(0.002)	(0.004)
Number of id	7,345	8,518
Observations	28,056	41,767

Table A1: Estimation of the propensity scores

Notes: Results from Probit regression. Dependent variable in column 1 takes the value one in the case of starter (a starter is a firm that for the first time in our sample period obtain technology transfers from universities). In column 1 the sample is restricted to firms that are either starters or that never have technology transfers over our sample period. In column 2, we do not restrict the sample and the dependent variable takes the value one in the case of technology transfers from universities. All regressors are lagged one year. Time and industry fixed effects are included in the regression. Robust standard errors in parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table A2: Effect of technology transfers from universities on firms' innovation with longer pre-treatment trend. Average treatment effect on the treated after matching with CDiD methodology

Dependent variable	Product innovation	Process innovation	Patents
	(1)	(2)	(3)
University technology transfers ^{CDiD}	0.045**	0.038	0.066***
	(0.026)	(0.027)	(0.021)
Observations	6,032	6,032	6,032
Number of id	3,322	3,322	3,322

Notes: University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table A3: Effect of technology transfers from universities on continuous measures of firm innovativeness. Average treatment effect on the treated after matching with CDiD methodology

Dependent variable: University technology transfers ^{CDiD}	Sales from not	1	Number of patents
	the market the firm		
	(1)	(2)	(3)
University technology transfers ^{CDiD}	0.534	0.763*	0.087**
	(0.406)	(0.402)	(0.031)
Observations	5,841	5,841	5,841
Number of id	3,453	3,453	3,453

Notes: University technology transfers^{CDiD} is the average treatment effect on the treated using the CDiD methodology. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table A4: Placebo test: Random assignment of university technology transfers. Average treatment effect on the treated after matching with CDiD methodology

Dependent variable	Product innovation	Process innovation	Patents
	(1)	(2)	(3)
Random university technology transfers ^{CDiD}	-0.009	-0.014	-0.011
	(0.030)	(0.031)	(0.023)
Observations	3,573	3,573	3,573
Number of id	2,542	2,542	2,542

Notes: For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Standard errors between parentheses. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

Table A5: Effect of technology	transfers from	universities on firms	s' innovation. I	V specification

Dependent variable	Product innovation	Process innovation	Patents
	(1)	(2)	(3)
University technology transfers	0.523***	0.276**	0.529***
	(0.153)	(0.117)	(0.148)
Observations	33,752	33,752	33,752
Number of id	7,290	7,290	7,290
R-squared	0.335	0.474	0.345
First stage results:			
Importance of conferences	0.165***	0.163***	0.150***
	(0.031)	(0.0331	(0.031)
Kleibergen-Paap F-Statistic	27.87	27.89	23.47

Notes: University technology transfers is a dummy variable that takes the value one if a firm has expenditures in R&D services from Spanish universities. Sector x year and year FEs in all regressions. For exact definitions and sources of all variables see Tables 1 and 2 and the main text. Estimated robust standard errors clustered at the firm level are in parentheses. 2SLS regressions. University technology transfers is instrumented using pre-sample values of the importance of conferences, expositions or trade fairs measured at the average of the industry and regional level. The F-statistics are reported for the Kleibergen–Paap test for weak identification. Estimations include initial values and one-year lag of the dependent variable. * Significant at 10%; ** Significant at 5%; *** significant at 1%.

ON-LINE APPENDIX

Do University Technology Transfers Increase Firms' Innovation?

Maria Garcia-Vega and Oscar Vicente-Chirivella

Table of Contents

Table OA1: Number of observations and percentages by sector of activity

Table OA2: Descriptive statistics distinguishing for the matched sample corresponding to Table 4

Table OA3: Balancing property for Table 4

Table OA4: Descriptive statistics distinguishing for the matched sample corresponding to Table 5

Table OA5: Balancing property for Table 5

Table OA6: Balancing property for table with longer pre-treatment

Table OA7: Balancing property for table for placebo test

Table OA8: Balancing property for companies with technology transfers from universities (treated) and companies with technology transfers from other providers (control)

Table OA9: Balancing property for heterogeneous effects: SMEs and non-SMEs

Table OA10: Balancing property for heterogeneous effects: Non-recession and recession

Table OA11: Balancing property for heterogeneous effects: Differentiation by Pavitt sectoral taxonomy

Table OA12: Balancing property for companies without technology transfers from universities located in regions and sectors with high technology transfers from universities (treated) and companies without technology transfers from universities located in regions and sectors with low technology transfers from universities (control)

Table OA1: Number of observations and percentages by sector of activity

	Technology transfers from Universities	Observations per sector	(1)/(2)	% over total observations with techno. transfers
	(1)	(2)	(3)	(4)
Agriculture	218	1,500	14.5%	2.8%
Mining and extractive industries	571	6,564	8.7%	7.4%
Food and tobacco	660	7,687	8.6%	8.6%
Textiles, printing and wood	262	10,552	2.5%	3.4%
Chemicals	678	5,995	11.3%	8.8%
Pharmaceuticals	560	1,646	34.0%	7.3%
Manufacturing of non-metallic products	333	7,053	4.7%	4.3%
Manufacturing of basic metals	641	10,846	5.9%	8.3%
Manufacturing of electrical and optimal equipment	933	10,714	8.7%	12.1%
Manufacturing of transport equipment	193	3,557	5.4%	2.5%
Wholesale and retail trade	292	9,122	3.2%	3.8%
Transport, storage and communication	350	7,911	4.4%	4.5%
Final intermediation	19	2,286	0.8%	0.2%
Real estate, renting and business activities	431	8,384	5.1%	5.6%
R&D services, software and technical analysis	1,328	9,677	13.7%	17.2%
Other services	245	9478	2.6%	3.2%
Total	7,714	112,972		

Notes: Column (1) shows the number of observations within a sector with technology transfers from Universities. Column (2) shows the number of observations per sector. Column (3) shows the ratio between columns (1) and (2). Column (4) shows the percentage of observations with technology transfers.

		Full sam	ple	Witl	h technology	transfers	Withou	it technology	transfers
Variable	Mean	Std. Dev.	Observations	Mean	Std. Dev.	Observations	Mean	Std. Dev.	Observations
Treatment variable									
Technology transfers from universities	0.505	0.500	1523	1.000	0.000	769	0.000	0.000	754
Outcome variables									
Product innovation	0.754	0.431	1523	0.769	0.422	769	0.740	0.439	754
Process innovation	0.721	0.449	1523	0.735	0.442	769	0.707	0.455	754
Patents	0.196	0.397	1523	0.222	0.416	769	0.170	0.376	754
Sales from products new to the market	6.884	7.223	1523	7.145	7.246	769	6.618	7.195	754
Sales from products new to the firm	7.885	7.217	1523	8.102	7.205	769	7.664	7.227	754
Number of patents	0.239	0.571	1523	0.276	0.615	769	0.201	0.520	754
Control variables									
Employment	4.147	1.552	1523	4.215	1.524	769	4.078	1.579	754
Physical investment	7.992	2.258	1391	8.147	2.218	721	7.826	2.291	670
Internal R&D expenditures	8.089	1.392	1382	8.145	1.335	741	8.025	1.453	641
Innovation expenditures	8.381	1.457	1459	8.502	1.379	769	8.246	1.529	690
Sales	11.487	2.045	1523	11.535	2.070	769	11.437	2.020	754
Exports	0.707	0.455	1523	0.724	0.447	769	0.690	0.463	754
Group	0.410	0.492	1523	0.421	0.494	769	0.399	0.490	754

Table OA2: Descriptive statistics for the matched sample corresponding to Table 4

Note: The sample is restricted to firms that are either starters or that never have technology transfers over our sample period. For exact definitions and sources of all variables see Tables 1 and 2 and the main text.

	Unmatched (U)	Me	ean	% bias	%reduct	t-	-test
Variable	Matched (M)	Treated	Control	-	bias	t	p-value
Product innovation	U	0.75	0.47	58.80		16.79	0.00
	Μ	0.74	0.75	-3.20	94.50	-0.69	0.49
Process innovation	U	0.71	0.49	45.50		13.25	0.00
	Μ	0.69	0.71	-4.00	91.30	-0.83	0.41
Patents	U	0.22	0.10	33.40		12.16	0.00
	Μ	0.19	0.19	0.70	97.90	0.13	0.90
Employment	U	4.28	4.10	10.60		3.12	0.00
	Μ	4.16	4.09	4.50	58.10	0.94	0.35
Capital investment	U	8.25	7.82	18.10		5.34	0.00
	Μ	8.06	7.96	4.10	77.40	0.87	0.38
Researchers	U	3.65	3.65	-0.80		-0.25	0.80
	Μ	3.62	3.62	0.20	78.60	0.03	0.97
Internal R&D	U	8.27	7.73	35.20		10.44	0.00
	Μ	7.99	7.97	1.70	95.30	0.36	0.72
Group	U	0.42	0.39	5.60		1.72	0.09
	Μ	0.39	0.39	0.00	100.00	0.00	1.00
Exports	U	0.71	0.49	47.90		13.90	0.00
		0.70	0.69	3.20	93.30	0.66	0.51

Table OA3: Balancing property for Table 4

Notes: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms. Results from Probit regression. The matched sample corresponds to starters (a starter is a firm that for the first time in our sample period obtain technology transfers from universities). The sample is restricted to firms that are either starters or that never have technology transfers over our sample period. The matching procedure is a one-to-one matching with replacement with caliper equal to 0.0001.

	Ps R2	LR chi2	p>chi2	MeanBias	MedBias
Unmatched	0.060	525.94	0.00	20.5	18.0
Matched	0.004	8.9	1.00	1.9	1.4

			1 st ma	tching				2 nd matching					3 rd matching					
		h techno transfer			out tech transfer	05		n techno transfer	. .		out techi transfer	05		h techno transfer			out techt transfer	05
Variable	М	SD	Obs	М	SD	Obs	М	SD	Obs	М	SD	Obs	М	SD	Obs	М	SD	Obs
Treatment variable																		
Technology transfers from universities	1.00	0.00	2518	0.00	0.00	2,462	0.00	0.00	833	0.00	0.00	817	0.00	0.00	1,521	0.00	0.00	1,475
Outcome variables																		
Product innovation	0.77	0.42	2518	0.76	0.43	2,462	0.78	0.42	833	0.78	0.41	817	0.78	0.42	1,521	0.78	0.41	1,475
Process innovation	0.72	0.45	2518	0.71	0.45	2,462	0.75	0.43	833	0.73	0.44	817	0.72	0.45	1,521	0.72	0.45	1,475
Patents	0.22	0.42	2518	0.19	0.39	2,462	0.22	0.41	833	0.21	0.41	817	0.21	0.40	1,521	0.17	0.38	1,475
Sales from products new to the market	7.67	7.17	2518	6.99	7.26	2,462	7.26	7.24	833	7.27	7.22	817	7.15	7.14	1,521	6.98	7.17	1,475
Sales from products new to the firm	7.90	7.22	2518	8.14	7.23	2,462	8.26	7.22	833	8.35	7.31	817	8.24	7.17	1,521	8.32	7.10	1,475
Number of patents	0.28	0.62	2518	0.24	0.61	2,462	0.26	0.58	833	0.26	0.59	817	0.26	0.62	1,521	0.22	0.55	1,475
Control variables																		
Employment	4.18	1.52	2518	4.20	1.49	2,462	4.26	1.51	833	4.20	1.52	817	4.17	1.45	1,521	4.12	1.47	1,475
Physical investment	8.05	2.32	2346	8.03	2.32	2,163	8.10	2.28	833	8.10	2.32	726	7.96	2.23	1,521	7.88	2.30	1,308
Internal R&D expenditures	8.22	1.34	2461	8.10	1.45	2,222	8.26	1.40	833	8.17	1.40	733	8.02	1.29	1,521	7.92	1.38	1,281
Innovation expenditures	8.60	1.35	2518	8.31	1.52	2,324	8.59	1.45	833	8.40	1.46	758	8.29	1.33	1,521	8.12	1.47	1,364
Sales	11.50	2.10	2518	11.57	1.98	2,458	11.56	2.07	833	11.57	2.08	817	11.52	1.91	1,521	11.50	1.94	1,474
Exports	0.75	0.43	2518	0.77	0.42	2,462	0.74	0.44	833	0.72	0.45	817	0.76	0.43	1,521	0.76	0.43	1,475
Group	0.44	0.50	2518	0.44	0.50	2,462	0.44	0.50	833	0.44	0.50	817	0.42	0.49	1,521	0.40	0.49	1,475

Table OA4: Descriptive statistics for the matched samples corresponding to Table 5

Note: M means mean; SD means standard deviation; Obs means observations. For exact definitions and sources of all variables see Tables 1 and 2 and the main text.

			1 st mat	ching				2 ^{nc}	¹ matching				3 ^r	rd matching		
	Unmatched (U) Matched				t	-test				t-	test				t-	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value
Product innovation	U	0.78	0.46	71.00	41.28	0.00	0.79	0.47	70.20	20.29	0.00	0.79	0.47	70.50	26.95	0.00
	М	0.76	0.76	-0.60	-0.26	0.80	0.78	0.78	-0.80	-0.18	0.86	0.78	0.79	-2.60	-0.82	0.41
Process innovation	U	0.76	0.48	60.40	35.58	0.00	0.74	0.49	53.80	16.03	0.00	0.73	0.49	52.90	20.86	0.00
	М	0.73	0.73	-0.10	-0.03	0.98	0.73	0.73	-1.30	-0.27	0.78	0.72	0.73	-1.70	-0.52	0.61
Patents	U	0.31	0.09	56.50	45.82	0.00	0.30	0.10	50.10	20.15	0.00	0.23	0.10	35.20	17.70	0.00
	М	0.20	0.21	-3.60	-1.27	0.20	0.24	0.25	-3.70	-0.68	0.50	0.21	0.20	3.40	0.88	0.38
Employment	U	4.55	4.09	27.10	16.82	0.00	4.42	4.10	19.40	5.87	0.00	4.27	4.10	10.60	4.17	0.00
	М	4.17	4.19	-1.40	-0.56	0.58	4.23	4.20	1.70	0.39	0.70	4.17	4.14	1.70	0.51	0.61
Capital investment	U	8.57	7.78	32.10	20.04	0.00	8.45	7.81	26.20	8.12	0.00	8.16	7.82	14.80	5.91	0.00
	М	8.02	8.08	-2.30	-0.91	0.37	8.18	8.05	5.50	1.20	0.23	8.02	7.93	3.60	1.05	0.29
Researchers	U	3.64	3.65	-1.50	-0.95	0.34	3.63	3.65	-2.60	-0.83	0.41	3.52	3.66	-16.60	-7.03	0.00
	М	3.61	3.55	7.30	2.65	0.01	3.63	3.63	0.60	0.12	0.90	3.57	3.58	-1.40	-0.41	0.69
Internal R&D	U	8.73	7.65	69.50	43.23	0.00	8.48	7.72	47.90	15.12	0.00	8.09	7.73	24.20	9.58	0.00
	М	8.14	8.11	1.90	0.81	0.42	8.15	8.12	2.00	0.46	0.65	7.92	7.83	6.20	1.91	0.06
Group	U	0.50	0.39	23.20	14.79	0.00	0.46	0.39	12.70	4.06	0.00	0.43	0.39	7.40	3.10	0.00
	М	0.42	0.43	-1.00	-0.38	0.70	0.42	0.42	-1.20	-0.25	0.81	0.41	0.39	3.80	1.09	0.28
Exports	U	0.77	0.48	63.80	37.40	0.00	0.74	0.49	53.70	16.02	0.00	0.77	0.48	62.60	24.19	0.00
	М	0.74	0.75	-1.90	-0.75	0.46	0.72	0.72	0.00	0.00	1.00	0.76	0.76	0.00	0.00	1.00

Table OA5: Balancing property for Table 5

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms. In the first matching we obtain 2,518 treated firms and 2,462 control firms. In the second matching we obtain 1,009 treated firms and 641 control firms, and in the third matching we get 1,772 treated firms and 1,230 control firms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.125	3155.6	0.00	26.8	23.2
	Matched	0.002	18.4	0.91	1.3	0.8
2 nd matching	Unmatched	0.052	476.2	0.00	23.9	20.4
	Matched	0.004	8.5	1.00	1.2	0.8
3rd matching	Unmatched	0.037	521.78	0.00	21.8	18.2
	Matched	0.003	12.8	0.99	1.7	1.4

			1 st mat	ching				2 nd	matching				3 ^r	d matching		
	Unmatched (U) Matabad				t	-test				t	-test				t-	test
Variable	Matched (M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value
Product innovation	U	0.78	0.78	0.40	0.14	0.89	0.79	0.48	67.80	17.33	0.00	0.78	0.48	65.10	22.62	0.00
	М	0.78	0.78	0.60	0.25	0.81	0.79	0.79	-0.30	-0.06	0.95	0.77	0.78	-0.70	-0.19	0.85
Process innovation	U	0.74	0.75	-1.60	-0.57	0.57	0.75	0.50	54.40	14.22	0.00	0.73	0.50	47.80	17.11	0.00
	М	0.74	0.75	-3.10	-1.18	0.24	0.75	0.77	-3.70	-0.78	0.43	0.72	0.71	0.80	0.22	0.83
Patents	U	0.26	0.26	0.60	0.24	0.81	0.30	0.10	51.30	18.29	0.00	0.22	0.10	31.10	13.87	0.00
	М	0.26	0.27	-1.00	-0.39	0.70	0.29	0.29	-1.40	-0.23	0.82	0.20	0.18	7.90	1.86	0.06
Employment	U	4.38	4.39	-0.30	-0.10	0.92	4.42	4.12	18.10	4.90	0.00	4.26	4.12	9.10	3.25	0.00
	М	4.38	4.44	-3.70	-1.42	0.16	4.36	4.39	-1.80	-0.37	0.71	4.15	4.16	-1.20	-0.32	0.75
Capital investment	U	8.35	8.38	-1.20	-0.42	0.67	8.50	7.84	27.00	7.49	0.00	8.23	7.84	16.40	5.87	0.00
	М	8.35	8.46	-4.40	-1.68	0.09	8.41	8.33	3.10	0.64	0.53	8.06	8.15	-3.70	-1.01	0.31
Researchers	U	3.64	3.63	2.10	0.77	0.44	3.59	3.66	-8.10	-2.33	0.02	-0.87	-0.84	-4.40	-1.65	0.10
	М	3.64	3.63	1.40	0.54	0.59	3.59	3.59	0.20	0.05	0.96	-0.87	-0.85	-1.70	-0.44	0.66
Internal R&D	U	8.47	8.47	-0.30	-0.11	0.91	8.53	7.71	52.40	14.59	0.00	8.05	7.72	22.00	7.98	0.00
	М	8.47	8.55	-5.80	-2.19	0.03	8.42	8.42	0.00	0.01	1.00	7.88	7.85	2.00	0.57	0.57
Group	U	0.46	0.46	0.10	0.05	0.96	0.47	0.39	17.20	4.87	0.00	0.41	0.39	4.20	1.57	0.12
	М	0.46	0.48	-2.70	-1.04	0.30	0.46	0.49	-5.30	-1.03	0.30	0.39	0.41	-3.20	-0.80	0.42
Exports	U	0.74	0.73	2.60	0.95	0.34	0.75	0.49	56.00	14.67	0.00	0.74	0.49	53.00	18.87	0.00
	М	0.74	0.73	2.00	0.78	0.44	0.74	0.73	3.10	0.64	0.52	0.73	0.76	-4.90	-1.35	0.18

Table OA6: Balancing property for table with longer pre-treatment

M 0.74 0.73 2.00 0.78 0.44 0.74 0.73 3.10 0.64 0.52 0.73 0.76 -4.90 -1.35 0.18 Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in two lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.002	11.5	0.99	1.0	9.3
	Matched	0.003	22.7	0.7	1.6	12.5
2nd matching	Unmatched	0.059	430.4	0.00	26.1	23.6
	Matched	0.006	13.03	0.98	2.2	1.8
3rd matching	Unmatched	0.066	941.4	0.00	25.7	22.1
	Matched	0.003	14.41	0.98	1.5	1.0

			1 st ma	atching					2 nd matching	g			3 rd	matching		
	Unmatched (U) Matched				t-	test				t-1	test				t-t	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value
Product innovation	U	0.76	0.45	66.30	23.28	0.00	0.77	0.45	69.40	20.78	0.00	0.77	0.45	69.40	19.39	0.00
	М	0.76	0.77	-0.50	-0.14	0.89	0.78	0.77	2.40	0.59	0.56	0.78	0.76	4.20	0.95	0.34
Process innovation	U	0.73	0.47	54.80	19.54	0.00	0.73	0.47	54.90	16.78	0.00	0.74	0.47	56.50	16.09	0.00
	М	0.73	0.73	0.00	0.00	1.00	0.73	0.71	5.50	1.30	0.20	0.74	0.73	1.00	0.21	0.83
Patents	U	0.18	0.09	26.40	11.69	0.00	0.22	0.09	35.80	14.29	0.00	0.19	0.09	29.50	10.68	0.00
	М	0.17	0.17	-0.90	-0.20	0.84	0.20	0.20	0.60	0.11	0.91	0.19	0.21	-7.50	-1.36	0.17
Employment	U	4.16	4.08	4.70	1.67	0.09	4.18	4.08	6.30	1.91	0.06	4.19	4.08	6.80	1.87	0.06
	Μ	4.12	4.13	-0.60	-0.19	0.85	4.13	4.14	-0.70	-0.18	0.86	4.16	4.18	-1.00	-0.23	0.82
Capital investment	U	7.83	7.76	2.80	1.05	0.30	7.99	7.76	9.80	3.05	0.00	7.94	7.76	7.70	2.24	0.03
	Μ	7.79	7.81	-0.80	-0.22	0.82	7.92	7.99	-3.10	-0.72	0.47	7.90	7.95	-2.20	-0.48	0.63
Researchers	U	3.63	3.65	-3.20	-1.23	0.22	3.61	3.65	-5.80	-1.89	0.06	3.56	3.65	-11.90	-3.57	0.00
	Μ	3.63	3.59	4.30	1.16	0.25	3.60	3.61	-1.90	-0.42	0.67	3.57	3.57	-0.50	-0.11	0.92
Internal R&D	U	7.77	7.61	10.80	4.02	0.00	7.87	7.61	17.00	5.46	0.00	7.85	7.61	16.00	4.68	0.00
	Μ	7.73	7.74	-0.20	-0.05	0.96	7.79	7.76	1.60	0.37	0.71	7.80	7.84	-3.10	-0.64	0.52
Group	U	0.41	0.39	4.10	1.54	0.12	0.39	0.39	1.20	0.40	0.69	0.40	0.39	1.90	0.58	0.56
	Μ	0.40	0.40	-1.00	-0.27	0.79	0.39	0.37	2.70	0.60	0.55	0.39	0.38	1.60	0.34	0.73
Exports	U	0.73	0.47	55.70	19.82	0.00	0.73	0.47	54.60	16.71	0.00	0.79	0.47	68.80	19.02	0.00
	М	0.74	0.76	-4.70	-1.32	0.19	0.73	0.72	2.30	0.55	0.58	0.78	0.80	-3.90	-0.93	0.35

Table OA7: Balancing property for table for placebo test

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.026	296.51	0.00	17.4	10.8
	Matched	0.003	11.55	0.99	1.5	0.6
2 nd matching	Unmatched	0.026	237.4	0.00	19.2	13.8
	Matched	0.005	14.79	0.98	2.0	1.9
3rd matching	Unmatched	0.028	233.6	0.00	20.5	16.0
	Matched	0.005	11.33	0.99	1.9	1.3

			1 st ma	atching					2nd matching				3 rd	matching		
	Unmatched (U) Matched				t·	-test				t-	test				t-	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value
Product innovation	U	0.78	0.72	15.80	8.91	0.00	0.83	0.73	24.30	5.11	0.00	0.82	0.73	21.70	5.20	0.00
	М	0.79	0.80	-0.70	-0.19	0.85	0.84	0.76	20.10	2.17	0.03	0.86	0.83	6.70	0.87	0.39
Process innovation	U	0.76	0.69	16.00	9.04	0.00	0.77	0.70	16.50	3.58	0.00	0.76	0.70	14.50	3.57	0.00
	Μ	0.73	0.75	-3.50	-0.93	0.36	0.75	0.75	0.00	0.00	1.00	0.73	0.75	-4.50	-0.51	0.61
Patents	U	0.31	0.20	25.50	15.53	0.00	0.33	0.22	24.00	5.76	0.00	0.26	0.22	8.80	2.31	0.02
	Μ	0.24	0.25	-3.80	-0.99	0.32	0.23	0.26	-7.30	-0.78	0.44	0.26	0.24	4.60	0.51	0.61
Employment	U	4.55	4.35	11.90	6.95	0.00	4.64	4.38	16.40	3.62	0.00	4.45	4.38	4.20	1.07	0.28
	М	4.25	4.29	-2.50	-0.66	0.51	4.35	4.47	-7.10	-0.78	0.44	4.22	4.25	-1.80	-0.24	0.81
Capital investment	U	8.57	8.32	10.30	5.90	0.00	8.72	8.36	15.00	3.34	0.00	8.51	8.36	6.20	1.55	0.12
	М	8.22	8.22	-0.10	-0.02	0.98	8.38	8.53	-6.40	-0.68	0.50	8.23	8.25	-0.90	-0.11	0.91
Researchers	U	3.64	3.57	8.50	4.81	0.00	3.59	3.59	0.50	0.12	0.91	3.58	3.59	-0.70	-0.17	0.87
	М	3.58	3.58	-0.40	-0.10	0.92	3.58	3.61	-3.10	-0.33	0.74	3.60	3.50	11.50	1.28	0.20
Internal R&D	U	8.73	8.15	37.00	21.18	0.00	8.77	8.27	32.60	7.23	0.00	8.37	8.28	5.90	1.48	0.14
	М	8.29	8.24	3.30	0.95	0.34	8.36	8.31	2.70	0.31	0.76	8.27	8.31	-2.60	-0.31	0.76
Group	U	0.50	0.49	3.50	2.05	0.04	0.52	0.49	6.90	1.55	0.12	0.49	0.49	0.60	0.16	0.87
	Μ	0.45	0.46	-1.60	-0.43	0.67	0.49	0.51	-3.70	-0.38	0.70	0.46	0.51	-10.30	-1.16	0.25
Exports	U	0.77	0.71	15.20	8.66	0.00	0.78	0.72	15.50	3.35	0.00	0.78	0.72	14.70	3.60	0.00
	М	0.74	0.75	-1.70	-0.44	0.66	0.79	0.80	-1.10	-0.12	0.91	0.79	0.77	3.70	0.43	0.67

Table OA8: Balancing property for companies with technology transfers from universities (treated) and companies with technology transfers from other providers (control)

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.115	1910.4	0.00	17.1	15.5
	Matched	0.004	14.2	0.97	1.6	21.4
2 nd matching	Unmatched	0.035	158.7	0.00	14.0	12.4
	Matched	0.032	19.1	0.86	4.2	2.7
3rd matching	Unmatched	0.032	171.23	0.00	10.8	8.0
	Matched	0.018	12.7	0.99	3.4	2.6

SMEs			1 st mat	ching				2 nd	matching				3 ^r	rd matching		
	Unmatched (U) Matched				t	-test				t-	-test				t-	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value
Product innovation	U	0.77	0.50	59.60	29.56	0.00	0.77	0.51	57.80	14.53	0.00	0.80	0.51	65.30	21.02	0.00
	М	0.76	0.78	-4.00	-1.40	0.16	0.77	0.79	-4.90	-0.95	0.34	0.79	0.80	-3.10	-0.83	0.41
Process innovation	U	0.72	0.49	47.60	24.33	0.00	0.69	0.50	39.10	10.27	0.00	0.72	0.50	46.70	15.76	0.00
	М	0.70	0.73	-5.30	-1.75	0.08	0.68	0.68	1.30	0.24	0.81	0.71	0.71	1.70	0.41	0.68
Patents	U	0.28	0.10	46.10	30.57	0.00	0.27	0.11	43.40	14.67	0.00	0.21	0.10	28.90	11.95	0.00
	М	0.19	0.19	0.40	0.12	0.90	0.21	0.23	-5.90	-0.95	0.34	0.19	0.18	2.90	0.64	0.52
Employment	U	3.76	3.49	22.50	11.49	0.00	3.72	3.50	18.90	4.85	0.00	3.68	3.50	15.30	5.10	0.00
	М	3.63	3.63	-0.10	-0.03	0.98	3.63	3.62	1.30	0.25	0.80	3.64	3.61	2.40	0.63	0.53
Capital investment	U	7.71	7.15	27.30	13.98	0.00	7.66	7.17	23.80	6.25	0.00	7.52	7.17	16.80	5.75	0.00
	М	7.41	7.39	1.10	0.35	0.72	7.49	7.46	1.30	0.25	0.81	7.45	7.50	-2.30	-0.58	0.57
Researchers	U	3.71	3.71	0.60	0.31	0.75	3.74	3.71	3.90	1.05	0.29	3.60	3.71	-14.30	-5.08	0.00
	М	3.66	3.66	-0.70	-0.20	0.84	3.70	3.74	-5.10	-0.91	0.36	3.63	3.62	2.00	0.48	0.63
Internal R&D	U	8.30	7.40	65.60	34.59	0.00	8.11	7.45	46.80	12.83	0.00	7.74	7.46	21.80	7.23	0.00
	М	7.79	7.77	1.90	0.73	0.47	7.83	7.76	4.60	0.90	0.37	7.65	7.59	4.30	1.09	0.28
Group	U	0.36	0.31	10.90	5.95	0.00	0.34	0.31	6.20	1.71	0.09	0.34	0.31	7.80	2.80	0.01
	М	0.34	0.34	0.40	0.13	0.89	0.32	0.32	-0.70	-0.12	0.90	0.32	0.31	1.30	0.31	0.75
Exports	U	0.75	0.58	37.50	18.94	0.00	0.72	0.58	28.50	7.45	0.00	0.74	0.58	34.90	11.70	0.00
	М	0.72	0.74	-4.40	-1.47	0.14	0.70	0.74	-8.80	-1.64	0.10	0.73	0.73	0.00	0.00	1.00

Table OA9: Balancing property for heterogeneous effects: SMEs and non-SMEs

M 0.72 0.74 -4.40 -1.47 0.14 0.70 0.74 -8.80 -1.64 0.10 0.73 0.73 0.00 0.00 1.00 Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.107	2018	0.00	23.6	16.5
	Matched	0.003	18.9	0.90	1.3	0.7
2 nd matching	Unmatched	0.052	360.7	0.00	20.6	17.2
	Matched	0.009	15.2	0.97	2.6	1.3
3rd matching	Unmatched	0.035	362.4	0.00	18.9	14.9
	Matched	0.003	10.4	0.99	1.5	1.5

Non-SMEs			1 st mat	ching					2nd matching				3 rd n	natching		
	Unmatched (U) Matched				t-	test				t-te	est				t-	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value	Treated	Control	% bias	t	p- value
Product innovation	U	0.81	0.38	97.40	29.47	0.00	0.84	0.40	100.40	13.69	0.00	0.82	0.40	94.80	9.21	0.00
	М	0.74	0.79	-11.40	-1.13	0.26	0.82	0.82	0.00	0.00	1.00	0.83	0.83	0.00	0.00	1.00
Process innovation	U	0.86	0.49	85.60	24.66	0.00	0.90	0.51	94.90	12.06	0.00	0.86	0.51	82.30	7.64	0.00
	Μ	0.83	0.82	1.30	0.14	0.89	0.88	0.87	3.90	0.47	0.64	0.86	0.88	-4.90	-0.43	0.66
Patents	U	0.42	0.09	80.80	35.81	0.00	0.37	0.10	66.70	13.47	0.00	0.25	0.11	37.70	4.96	0.00
	М	0.22	0.34	-27.70	-2.37	0.02	0.32	0.32	0.00	0.00	1.00	0.26	0.23	8.40	0.50	0.62
Employment	U	6.59	6.46	13.60	4.69	0.00	6.51	6.46	5.70	0.84	0.40	6.43	6.46	-3.90	-0.43	0.67
	М	6.46	6.36	10.30	0.96	0.34	6.46	6.48	-2.10	-0.20	0.84	6.43	6.50	-6.70	-0.46	0.64
Capital investment	U	10.80	9.79	45.10	14.57	0.00	10.82	9.84	46.30	6.58	0.00	10.66	9.85	39.90	3.84	0.00
	М	10.64	10.33	13.80	1.53	0.13	10.61	10.53	3.60	0.38	0.70	10.65	10.28	17.70	1.44	0.15
Researchers	U	3.45	3.35	10.90	3.45	0.00	3.33	3.36	-3.50	-0.53	0.59	3.18	3.36	-18.60	-1.99	0.05
	Μ	3.36	3.29	6.90	0.69	0.49	3.27	3.30	-2.90	-0.28	0.78	3.25	3.12	14.00	0.91	0.37
Internal R&D	U	9.86	8.96	55.50	16.71	0.00	9.59	9.07	32.00	4.61	0.00	9.38	9.08	19.30	1.89	0.06
	М	9.39	9.18	12.90	1.32	0.19	9.37	9.25	7.50	0.78	0.44	9.24	9.34	-6.30	-0.45	0.65
Group	U	0.88	0.70	46.00	13.28	0.00	0.81	0.71	22.40	3.24	0.00	0.79	0.71	19.30	1.98	0.05
	М	0.85	0.82	9.90	1.00	0.32	0.83	0.85	-5.00	-0.56	0.57	0.79	0.78	2.40	0.18	0.86
Exports	U	0.83	0.52	69.70	20.60	0.00	0.79	0.54	56.90	7.97	0.00	0.79	0.54	56.80	5.59	0.00
	М	0.80	0.78	6.30	0.65	0.52	0.81	0.81	-1.20	-0.13	0.90	0.81	0.82	-2.30	-0.19	0.85

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.175	1055.3	0.00	33.3	21.3
	Matched	0.055	27.05	0.46	6.3	3.3
2 nd matching	Unmatched	0.063	128	0.00	29.4	22.4
	Matched	0.014	7.5	1.00	2.8	2.1
3rd matching	Unmatched	0.061	70.34	0.00	27.2	19.3
	Matched	0.045	11.8	0.99	5.7	3.4

Non-Recession			1 st ma	atching					2nd matching				3 rd	1 matching		
	Unmatched (U) Matched				t-	test				t-t	est				t-t	est p-
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	value
Product innovation	U	0.77	0.46	67.00	23.66	0.00	0.76	0.47	60.80	10.91	0.00	0.76	0.47	62.40	14.72	0.00
	М	0.75	0.75	-0.60	-0.15	0.88	0.75	0.74	2.00	0.27	0.79	0.75	0.76	-2.20	-0.41	0.68
Process innovation	U	0.75	0.47	58.40	20.95	0.00	0.68	0.48	42.20	7.84	0.00	0.72	0.48	51.80	12.48	0.00
	М	0.71	0.72	-2.30	-0.56	0.58	0.67	0.70	-5.10	-0.68	0.50	0.70	0.74	-6.50	-1.17	0.24
Patents	U	0.29	0.11	46.70	21.72	0.00	0.28	0.12	42.70	9.99	0.00	0.26	0.11	38.30	11.60	0.00
	М	0.21	0.22	-3.00	-0.67	0.50	0.24	0.22	6.30	0.74	0.46	0.23	0.21	4.50	0.71	0.48
Employment	U	4.26	4.14	6.90	2.62	0.01	4.24	4.15	5.50	1.03	0.30	4.10	4.15	-3.00	-0.71	0.48
	М	3.99	3.99	0.20	0.05	0.96	4.10	4.04	3.90	0.53	0.60	3.99	3.89	5.70	1.07	0.28
Capital investment	U	8.45	7.86	24.20	9.09	0.00	8.21	7.89	13.40	2.53	0.01	8.19	7.89	12.90	3.14	0.00
	М	8.01	8.16	-6.00	-1.53	0.13	8.06	7.96	4.20	0.55	0.58	8.00	7.81	8.00	1.50	0.13
Researchers	U	3.65	3.68	-3.80	-1.46	0.15	3.67	3.68	-2.10	-0.40	0.69	3.56	3.69	-16.20	-4.09	0.00
	М	3.62	3.61	1.30	0.30	0.77	3.68	3.65	3.40	0.43	0.67	3.58	3.64	-7.90	-1.37	0.17
Internal R&D	U	8.42	7.55	56.20	21.21	0.00	8.23	7.59	40.10	7.82	0.00	7.91	7.59	20.60	5.01	0.00
	М	7.96	7.94	1.50	0.41	0.69	7.96	7.82	8.60	1.19	0.23	7.74	7.65	5.90	1.09	0.28
Group	U	0.43	0.38	10.40	3.98	0.00	0.41	0.38	5.80	1.11	0.27	0.36	0.38	-4.40	-1.12	0.26
	М	0.37	0.35	4.60	1.10	0.27	0.37	0.35	4.40	0.57	0.57	0.35	0.31	8.20	1.43	0.15
Exports	U	0.74	0.48	56.10	20.22	0.00	0.71	0.48	47.50	8.73	0.00	0.74	0.48	54.70	13.07	0.00
	М	0.70	0.72	-4.40	-1.07	0.29	0.68	0.66	5.80	0.75	0.45	0.72	0.72	0.70	0.13	0.90

Table OA10: Balancing property for heterogeneous effects: Non-recession and recession

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.154	1475	0.00	30.8	24.2
	Matched	0.005	14.2	0.90	2.5	1.9
2nd matching	Unmatched	0.087	295.9	0.00	28.3	20.9
	Matched	0.011	9.8	0.98	3.6	3.6
3rd matching	Unmatched	0.086	445.8	0.00	27.9	21.7
	Matched	0.013	20.2	0.57	4.1	4.1

Recession			1 st ma	atching					2 nd matching	5			3 rd	matching		
	Unmatched (U) Matched				t-	-test				t-1	test				t-1	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value
Product innovation	U	0.80	0.46	76.30	30.44	0.00	0.83	0.47	81.10	16.04	0.00	0.81	0.47	76.90	19.81	0.00
	М	0.78	0.77	2.10	0.55	0.58	0.81	0.81	1.10	0.18	0.86	0.80	0.81	-1.20	-0.26	0.79
Process innovation	U	0.78	0.49	63.20	25.67	0.00	0.78	0.49	61.90	12.70	0.00	0.75	0.49	55.90	14.96	0.00
	М	0.75	0.75	1.10	0.29	0.78	0.77	0.75	3.20	0.49	0.62	0.74	0.73	3.50	0.71	0.48
Patents	U	0.33	0.09	63.40	37.41	0.00	0.31	0.09	55.40	16.29	0.00	0.21	0.09	34.20	11.93	0.00
	М	0.19	0.22	-5.90	-1.39	0.17	0.23	0.27	-10.10	-1.28	0.20	0.19	0.18	3.80	0.67	0.51
Employment	U	4.74	4.05	41.10	17.49	0.00	4.53	4.07	28.00	5.85	0.00	4.39	4.07	20.10	5.33	0.00
	Μ	4.32	4.37	-3.00	-0.83	0.41	4.30	4.23	3.80	0.60	0.55	4.30	4.32	-1.60	-0.34	0.74
Capital investment	U	8.59	7.73	34.70	15.23	0.00	8.58	7.76	33.60	7.35	0.00	8.15	7.76	16.20	4.44	0.00
	М	7.99	8.00	-0.60	-0.16	0.87	8.22	7.97	10.60	1.60	0.11	8.03	8.00	1.40	0.29	0.78
Researchers	U	3.63	3.63	0.30	0.13	0.90	3.62	3.63	-1.80	-0.40	0.69	3.49	3.63	-17.10	-5.07	0.00
	Μ	3.61	3.50	13.40	3.16	0.00	3.60	3.62	-3.00	-0.44	0.66	3.54	3.54	0.10	0.01	0.99
Internal R&D	U	8.97	7.76	78.40	33.65	0.00	8.63	7.83	51.60	11.34	0.00	8.23	7.84	26.40	7.11	0.00
	М	8.31	8.26	3.30	0.94	0.35	8.29	8.30	-1.20	-0.19	0.85	8.05	7.96	6.20	1.30	0.19
Group	U	0.55	0.40	30.10	13.34	0.00	0.48	0.40	15.00	3.36	0.00	0.48	0.40	14.90	4.29	0.00
	М	0.46	0.50	-7.60	-1.82	0.07	0.44	0.45	-3.00	-0.42	0.67	0.45	0.45	-1.10	-0.21	0.84
Exports	U	0.80	0.48	70.90	28.37	0.00	0.76	0.49	59.20	12.25	0.00	0.82	0.49	74.30	19.05	0.00
	М	0.78	0.80	-4.50	-1.22	0.22	0.75	0.77	-3.20	-0.49	0.62	0.80	0.81	-0.90	-0.20	0.84

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.107	1248.7	0.00	31.8	25.3
	Matched	0.008	27.5	0.28	2.5	2.2
2 nd matching	Unmatched	0.034	151.1	0.00	27.8	21.4
	Matched	0.009	10.0	0.99	2.5	1.8
3rd matching	Unmatched	0.013	88.2	0.00	25.5	18.9
	Matched	0.013	26.4	0.33	2.6	1.1

Table OA11: Balancing property for Pavitt sectoral taxonomy

Table OA11a: Science based sectors

			1 st m	atching					2 nd matching				3 ^r	d matching		
	Unmatched (U) Matched				t-	test				t-	test				t	-test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value
Product innovation	U	0.83	0.65	43.00	16.01	0.00	0.82	0.66	36.40	6.85	0.00	0.81	0.66	34.60	6.47	0.00
	М	0.81	0.83	-3.90	-0.79	0.43	0.79	0.84	-9.80	-1.24	0.22	0.81	0.83	-5.90	-0.83	0.40
Process innovation	U	0.72	0.53	39.10	15.32	0.00	0.68	0.55	26.90	5.37	0.00	0.69	0.55	29.90	5.87	0.00
	М	0.66	0.67	-3.20	-0.58	0.56	0.66	0.67	-0.80	-0.09	0.93	0.65	0.68	-5.40	-0.68	0.50
Patents	U	0.37	0.14	55.10	26.42	0.00	0.31	0.15	37.30	8.78	0.00	0.16	0.15	2.50	0.51	0.61
	М	0.16	0.20	-9.40	-1.87	0.06	0.17	0.19	-5.60	-0.68	0.50	0.14	0.15	-2.60	-0.34	0.73
Employment	U	4.21	3.56	43.40	17.37	0.00	4.07	3.60	30.90	6.19	0.00	3.78	3.61	11.70	2.24	0.03
	М	3.70	3.68	1.30	0.25	0.80	3.75	3.68	4.20	0.53	0.60	3.66	3.55	7.30	0.99	0.32
Capital investment	U	8.18	7.14	45.30	17.85	0.00	7.95	7.24	30.80	6.16	0.00	7.50	7.25	11.30	2.19	0.03
	М	7.35	7.29	2.40	0.46	0.64	7.39	7.47	-3.30	-0.41	0.68	7.23	7.34	-4.90	-0.68	0.50
Researchers	U	3.80	3.76	4.50	1.77	0.08	3.77	3.77	1.10	0.22	0.82	3.70	3.77	-9.00	-1.82	0.07
	М	3.74	3.77	-3.60	-0.65	0.52	3.73	3.70	3.20	0.35	0.73	3.70	3.72	-1.30	-0.16	0.87
Internal R&D	U	9.08	7.97	69.00	27.57	0.00	8.74	8.08	40.80	8.27	0.00	8.15	8.09	3.50	0.67	0.51
	М	8.26	8.14	7.80	1.75	0.08	8.15	8.21	-3.60	-0.47	0.64	8.02	8.00	1.40	0.18	0.85
Group	U	0.44	0.37	14.50	5.97	0.00	0.39	0.37	4.00	0.84	0.40	0.34	0.38	-6.50	-1.31	0.19
	М	0.34	0.34	1.50	0.29	0.77	0.37	0.38	-3.10	-0.36	0.72	0.32	0.31	0.70	0.09	0.93
Exports	U	0.81	0.59	49.40	18.59	0.00	0.74	0.60	30.20	5.89	0.00	0.72	0.60	25.90	5.06	0.00
	М	0.71	0.71	0.30	0.06	0.95	0.69	0.69	0.00	0.00	1.00	0.71	0.70	2.10	0.26	0.79

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.125	1230.5	0.00	28.7	27.9
	Matched	0.01	12.2	0.87	2.5	1.5
2nd matching	Unmatched	0.05	179	0.00	20.8	18.8
	Matched	0.01	6.1	0.99	2.4	0.9
3rd matching	Unmatched	0.04	151.95	0.00	16.6	9.00
	Matched	0.01	5.40	0.99	1.5	0.00

Table OA11b: Specialized suppliers

			1 st ma	atching				:	2nd matching				3 rd	matching		
	Unmatched (U) Matched				t-	test				t	-test				t-	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value
Product innovation	U	0.88	0.63	62.20	10.57	0.00	0.92	0.64	70.90	5.97	0.00	0.88	0.63	60.00	7.37	0.00
	Μ	0.88	0.88	0.00	0.00	1.00	0.89	0.91	-6.30	-0.52	0.60	0.87	0.88	-1.40	-0.16	0.87
Process innovation	U	0.72	0.51	44.60	8.44	0.00	0.74	0.51	47.20	4.57	0.00	0.70	0.51	38.90	5.36	0.00
	М	0.72	0.74	-2.60	-0.35	0.72	0.70	0.76	-13.20	-0.89	0.38	0.72	0.82	-20.80	-2.19	0.03
Patents	U	0.39	0.18	47.90	10.75	0.00	0.46	0.18	61.90	7.30	0.00	0.29	0.19	25.10	3.91	0.00
	М	0.33	0.28	12.00	1.45	0.15	0.43	0.34	19.50	1.14	0.26	0.28	0.28	0.00	0.00	1.00
Employment	U	4.72	3.73	68.00	13.71	0.00	4.71	3.75	68.20	6.86	0.00	4.50	3.75	52.20	7.47	0.00
	Μ	4.43	4.39	2.30	0.32	0.75	4.57	4.31	18.60	1.32	0.19	4.31	4.39	-5.90	-0.58	0.56
Capital investment	U	8.56	7.43	52.60	10.12	0.00	8.44	7.47	45.00	4.48	0.00	8.15	7.47	32.00	4.45	0.00
	Μ	8.18	8.09	4.30	0.59	0.55	8.18	7.93	11.70	0.76	0.45	7.94	8.30	-17.10	-1.65	0.10
Researchers	U	3.49	3.54	-7.20	-1.37	0.17	3.54	3.54	0.20	0.02	0.98	3.40	3.55	-17.50	-2.60	0.01
	Μ	3.46	3.47	-2.40	-0.30	0.77	3.48	3.59	-13.50	-0.84	0.40	3.38	3.41	-3.80	-0.34	0.73
Internal R&D	U	8.78	7.66	74.00	14.55	0.00	8.86	7.70	76.10	7.76	0.00	8.32	7.70	43.10	5.79	0.00
	Μ	8.38	8.33	3.60	0.53	0.60	8.56	8.47	6.30	0.46	0.64	8.08	8.16	-5.70	-0.60	0.55
Group	U	0.56	0.33	45.90	9.34	0.00	0.51	0.34	35.00	3.69	0.00	0.53	0.34	39.90	5.88	0.00
	М	0.49	0.53	-8.40	-1.03	0.31	0.48	0.45	5.10	0.32	0.75	0.46	0.49	-4.80	-0.43	0.67
Exports	U	0.93	0.76	50.60	8.30	0.00	0.94	0.76	53.30	4.41	0.00	0.91	0.76	41.10	5.04	0.00
	М	0.92	0.93	-2.70	-0.45	0.65	0.94	0.95	-3.60	-0.34	0.73	0.91	0.94	-6.50	-0.81	0.42

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.09	253.4	0.00	28.7	21.8
	Matched	0.01	6.31	0.99	2.2	0.6
2nd matching	Unmatched	0.08	88.6	0.00	28.5	22.4
	Matched	0.04	9.8	0.93	5.8	3.6
3rd matching	Unmatched	0.42	73.8	0.00	24.9	21.5
	Matched	0.03	13.9	0.73	4.0	0.0

Table OA11c:	Scale and	information	intoncivo	contore
Table OATIC.	Scale allu	i mitor mation	mensive	3001013

			1 st ma	atching				,	2nd matching				3 rd	matching		
	Unmatched (U) Matched				t-	test				t·	test				t-t	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value
Product innovation	U	0.72	0.44	59.40	16.04	0.00	0.70	0.45	54.20	7.02	0.00	0.77	0.44	70.20	12.60	0.00
	М	0.73	0.74	-2.20	-0.40	0.69	0.71	0.75	-9.60	-0.90	0.37	0.76	0.79	-6.00	-0.85	0.40
Process innovation	U	0.83	0.55	64.80	16.32	0.00	0.82	0.55	60.10	7.26	0.00	0.85	0.55	69.80	11.76	0.00
	Μ	0.81	0.82	-2.30	-0.45	0.65	0.79	0.82	-7.30	-0.72	0.47	0.84	0.83	1.40	0.21	0.83
Patents	U	0.32	0.09	58.80	21.74	0.00	0.31	0.10	54.60	9.58	0.00	0.29	0.10	49.80	12.32	0.00
	Μ	0.24	0.27	-7.00	-1.08	0.28	0.25	0.30	-13.40	-1.02	0.31	0.25	0.32	-17.00	-1.84	0.07
Employment	U	5.49	4.50	58.60	17.63	0.00	5.25	4.52	45.80	6.14	0.00	5.15	4.52	41.10	7.65	0.00
	Μ	4.93	5.01	-4.70	-0.88	0.38	4.94	4.93	0.60	0.06	0.95	4.96	4.84	7.30	1.02	0.31
Capital investment	U	9.81	8.50	48.90	15.00	0.00	9.82	8.54	52.50	6.96	0.00	9.42	8.54	36.90	6.88	0.00
	Μ	8.99	9.11	-4.60	-0.88	0.38	9.37	9.17	8.30	0.85	0.40	9.20	8.91	11.90	1.66	0.10
Researchers	U	3.42	3.52	-11.40	-3.34	0.00	3.38	3.52	-15.50	-2.17	0.03	3.30	3.52	-25.60	-5.02	0.00
	Μ	3.40	3.38	2.10	0.36	0.72	3.42	3.46	-4.30	-0.39	0.70	3.36	3.36	-0.40	-0.06	0.95
Internal R&D	U	8.71	7.64	69.40	19.37	0.00	8.37	7.71	42.70	5.72	0.00	8.30	7.70	39.60	7.46	0.00
	Μ	8.22	8.21	0.00	0.00	1.00	8.07	8.17	-6.70	-0.65	0.51	8.09	7.96	8.80	1.22	0.22
Group	U	0.63	0.47	31.70	8.84	0.00	0.55	0.47	14.50	1.96	0.05	0.58	0.47	22.00	4.22	0.00
	Μ	0.55	0.58	-4.90	-0.82	0.41	0.50	0.56	-10.50	-0.91	0.36	0.56	0.53	6.20	0.79	0.43
Exports	U	0.71	0.58	27.30	7.43	0.00	0.72	0.58	30.10	3.88	0.00	0.80	0.58	50.00	8.80	0.00
	М	0.74	0.74	0.70	0.13	0.89	0.73	0.72	2.80	0.26	0.80	0.80	0.80	-0.70	-0.10	0.92

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.16	857	0.00	28.5	24.7
	Matched	0.01	7.5	0.99	1.9	0.7
2 nd matching	Unmatched	0.06	115.3	0.00	25.3	16.8
	Matched	0.03	12.	0.91	4.4	1.5
3rd matching	Unmatched	0.06	196	0.00	28.5	25.1
	Matched	0.01	11.0	0.96	3.2	0.7

			1 st ma	atching				-	2nd matching				3 rd	matching		
	Unmatched (U) Matched				t-	test				t-	test				t-1	test
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value
Product innovation	U	0.73	0.41	66.90	16.63	0.00	0.73	0.42	66.50	8.00	0.00	0.73	0.42	67.00	11.86	0.00
	М	0.72	0.72	-1.50	-0.26	0.79	0.76	0.71	9.20	0.81	0.42	0.73	0.66	14.70	1.86	0.06
Process innovation	U	0.83	0.49	75.70	17.52	0.00	0.81	0.50	69.40	7.88	0.00	0.77	0.50	58.80	10.12	0.00
	Μ	0.81	0.79	2.80	0.52	0.60	0.82	0.79	6.40	0.60	0.55	0.79	0.76	5.70	0.78	0.44
Patents	U	0.19	0.07	38.50	12.98	0.00	0.17	0.07	32.20	5.16	0.00	0.14	0.07	23.10	5.12	0.00
	Μ	0.15	0.14	2.70	0.42	0.68	0.14	0.18	-13.30	-0.98	0.33	0.14	0.17	-8.70	-0.90	0.37
Employment	U	4.44	4.21	14.90	3.79	0.00	4.51	4.21	19.80	2.35	0.02	4.31	4.21	6.40	1.15	0.25
	Μ	4.32	4.29	1.50	0.26	0.80	4.46	4.41	3.60	0.32	0.75	4.22	4.26	-2.60	-0.36	0.72
Capital investment	U	8.66	8.16	22.10	5.54	0.00	8.88	8.17	31.80	3.86	0.00	8.65	8.17	22.20	3.82	0.00
	Μ	8.49	8.58	-4.10	-0.73	0.47	8.82	8.66	7.00	0.64	0.52	8.51	8.33	8.30	1.10	0.27
Researchers	U	3.50	3.62	-16.00	-4.11	0.00	3.45	3.62	-21.80	-2.75	0.01	3.52	3.62	-12.10	-2.33	0.02
	Μ	3.53	3.49	5.90	0.97	0.33	3.49	3.53	-5.10	-0.40	0.69	3.59	3.54	6.50	0.83	0.41
Internal R&D	U	7.99	7.25	53.80	13.75	0.00	7.87	7.29	41.20	5.22	0.00	7.64	7.29	25.80	4.69	0.00
	М	7.73	7.71	1.50	0.26	0.80	7.80	7.66	9.70	0.87	0.39	7.49	7.49	0.10	0.01	0.99
Group	U	0.55	0.38	34.50	9.09	0.00	0.52	0.38	27.20	3.49	0.00	0.44	0.38	12.80	2.42	0.02
	Μ	0.50	0.49	2.90	0.47	0.64	0.50	0.49	1.50	0.12	0.91	0.41	0.42	-2.70	-0.33	0.74
Exports	U	0.80	0.65	34.30	8.26	0.00	0.81	0.65	35.30	4.10	0.00	0.82	0.65	39.00	6.61	0.00
	М	0.81	0.80	2.80	0.52	0.60	0.82	0.81	3.30	0.31	0.76	0.83	0.86	-6.80	-1.00	0.32

Table OA11d: Supplier dominated sectors

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.11	508	0.00	29.3	25.3
	Matched	0.00	4.2	0.99	1.3	0.4
2 nd matching	Unmatched	0.06	91.6	0.00	28	27.6
3rd matching	Matched	0.01	4.2	0.99	3.3	0.7
	Unmatched	0.04	127.1	0.00	23.3	19.7
	Matched	0.01	10.6	0.88	2.8	0.0

	1 st matching						2 nd matching				3 rd matching					
	Unmatched (U) Matched		t-test			t-test			test			t-test				
Variable	(M)	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p-value	Treated	Control	% bias	t	p- value
Product innovation	U	0.81	0.45	81.20	18.83	0.00	0.82	0.45	82.60	16.29	0.00	0.82	0.45	83.60	15.13	0.00
	М	0.82	0.82	0.40	0.08	0.94	0.81	0.81	0.00	0.00	1.00	0.82	0.84	-3.00	-0.49	0.63
Process innovation	U	0.72	0.47	51.30	12.63	0.00	0.71	0.47	50.60	10.65	0.00	0.73	0.47	53.70	10.34	0.00
	М	0.72	0.74	-3.30	-0.60	0.55	0.73	0.73	0.00	0.00	1.00	0.73	0.68	10.60	1.52	0.13
Patents	U	0.23	0.09	39.50	12.80	0.00	0.25	0.09	42.70	12.02	0.00	0.25	0.09	42.90	11.12	0.00
	М	0.22	0.23	-4.40	-0.64	0.52	0.23	0.23	-0.70	-0.08	0.94	0.23	0.29	-16.00	-1.82	0.07
Employment	U	4.12	4.08	2.60	0.58	0.56	4.21	4.08	8.90	1.67	0.10	4.24	4.08	10.80	1.92	0.06
	М	4.15	4.16	-0.70	-0.14	0.89	4.13	4.12	1.00	0.17	0.87	4.16	4.07	5.80	0.96	0.34
Capital investment	U	7.86	7.76	4.60	1.11	0.27	8.13	7.76	16.60	3.32	0.00	8.11	7.76	15.40	2.90	0.00
	М	7.88	7.92	-1.70	-0.32	0.75	7.99	7.96	1.10	0.17	0.87	7.99	7.92	3.50	0.52	0.60
Researchers	U	3.61	3.65	-5.80	-1.51	0.13	3.53	3.65	-14.50	-3.22	0.00	3.55	3.65	-12.40	-2.50	0.01
	М	3.57	3.55	2.20	0.37	0.72	3.53	3.57	-4.60	-0.69	0.49	3.57	3.59	-2.60	-0.38	0.71
Internal R&D	U	7.51	7.61	-7.60	-1.82	0.07	7.54	7.61	-5.00	-1.02	0.31	7.81	7.61	13.60	2.62	0.01
	М	7.63	7.71	-5.50	-1.02	0.31	7.59	7.55	2.50	0.39	0.70	7.71	7.70	1.00	0.15	0.88
Group	U	0.43	0.39	9.10	2.36	0.02	0.44	0.39	11.10	2.48	0.01	0.43	0.39	9.20	1.89	0.06
	М	0.42	0.43	-1.80	-0.30	0.76	0.42	0.40	4.40	0.63	0.53	0.41	0.41	1.60	0.22	0.82
Exports	U	0.83	0.47	81.30	18.57	0.00	0.86	0.47	89.60	17.07	0.00	0.88	0.47	97.10	16.63	0.00
	М	0.86	0.88	-5.60	-1.23	0.22	0.85	0.87	-5.00	-0.90	0.37	0.88	0.90	-4.40	-0.81	0.42

Table OA12: Balancing property for companies without technology transfers from universities located in regions and sectors with high technology transfers from universities (treated) and companies without technology transfers from universities located in regions and sectors with low technology transfers from universities (control)

Note: The table shows mean differences between treated and control observations for the matched sample based on the propensity score. All variables are in lags. The variables' employment, capital investment, researchers and internal R&D are in logarithms.

Overall balancing	tests					
0		Ps R2	LR chi2	p>chi2	MeanBias	MedBias
1st matching	Unmatched	0.15	966.4	0.00	29.0	25.1
2nd matching	Matched	0.00	6.8	1.00	1.8	0.7
	Unmatched	0.14	730	0.00	30.4	25.6
3rd matching	Matched	0.01	7.5	0.99	2.0	0.7
	Unmatched	0.09	414.4	0.00	30.9	27.3
	Matched	0.01	11.0	0.98	2.7	1.6